

COUNTY OF TULARE BOARD OF RETIREMENT

Leanne Malison Retirement Administrator

136 N AKERS STREET VISALIA, CALIFORNIA 93291 TELEPHONE (559) 713-2900 FAX (559) 730-2631 www.tcera.org

AGENDA OF THE BOARD OF RETIREMENT ADMINISTRATIVE COMMITTEE MEETING Wednesday, March 23, 2022 @ 10:00 a.m. TCERA Executive Room, 136 N. Akers Street, Visalia, CA 93291

NOTICE TO THE PUBLIC

Persons who wish to address the Board of Retirement during public comment or regarding an item that is on the agenda may address the Board of Retirement in person at the meeting. Any person addressing the Board will be limited to a maximum of five (5) minutes. A total of 15 minutes will be allotted for the Public Comment period unless otherwise extended by the Board Chair. If you are part of a large group that would like to comment on an agenda item, please consider commenting in writing or sending one spokesperson to speak on behalf of the group. Members of the public may also submit public comment via U.S. mail or via email to BORPublicComment@tcera.org before the meeting. The comments received via U.S. mail or email before the meeting will be read to the Board of Retirement in open session during the meeting as long as the comments meet the requirements for Public Comments as posted in the agenda. Public comments are limited to any item of interest to the public that is within the subject matter jurisdiction of the Board of Retirement. (Gov't Code Section 54954.3(a).)

Persons wishing to listen to the meeting may call the TCERA Office during regular business hours (Monday through Friday, 8:00 a.m. to 5:00 p.m.) within 48 hours of the meeting at 559-713-2900 for access information. Documents related to the items on this agenda are available for public inspection at the Retirement Office, 136 N Akers Street, Visalia, CA, during normal business hours. Such documents are also available on TCERA's website, www.tcera.org, subject to staff's ability to post the documents before the meeting.

As a courtesy to those in attendance, all individuals are requested to place cell phones and other electronic devices in the non-audible alert mode.

I. CALL TO ORDER

II. ROLL CALL

III. PUBLIC COMMENT

At this time, members of the public may comment on any item not appearing on the agenda. Under state law, matters presented under this item cannot be discussed or acted upon by the Board at this time. For items appearing on the agenda, the public is invited to make comments at the time the item comes up for Board consideration. Any person addressing the Board will be limited to a maximum of five (5) minutes so that all interested parties have an opportunity to speak. Please state your name for the record.

IV. AGENDA ITEMS

- 1. Review of Committee Minutes from the meeting of February 23, 2022.
- 2. Discussion and possible action regarding the following items:
 - a. TCERA and TCERA Property, Inc. Financial Statements December 31, 2021 and January 31, 2022
 - b. TCERA Disability Retirement Procedures Revision

V. UPCOMING MEETING

1. Wednesday, April 27, 2022 @ 10:00 a.m.

VI. ADJOURN

1. In compliance with the Americans with Disabilities Act, if you need special assistance to participate in this meeting, please contact the Secretary of the Board of Retirement at (559) 713-2900. Notification 48 hours prior to the meeting will help enable staff to make reasonable arrangements to ensure meaningful access. Documents related to the items on this Agenda submitted after distribution of the Agenda packet are available for public inspection at TCERA, 136 N. Akers Street, Visalia, CA. during normal business hours.



COUNTY OF TULARE **BOARD OF RETIREMENT**

Leanne Malison **Retirement Administrator**

136 N AKERS STREET VISALIA, CALIFORNIA 93291 TELEPHONE (559) 713-2900 FAX (559) 730-2631 www.tcera.org

MINUTES OF THE BOARD OF RETIREMENT ADMINISTRATIVE COMMITTEE MEETING Wednesday, February 23, 2022 @ 10:00 a.m. TCERA Executive Room, 136 N. Akers Street, Visalia, CA 93291

I. CALL TO ORDER

The meeting was called to order at 11:20 a.m.

II. ROLL CALL

Present: David Kehler, Chair; Cass Cook, Nathan Polk, Laura Hernandez George Finney (alternate), David Vasquez (alternate) Leanne Malison, Mary Warner

III. PUBLIC COMMENT

None

IV. AGENDA ITEMS

1. Review of Committee Minutes from the meeting November 17, 2021. Motion to approve minutes as presented. Motion: Hernandez Second: Polk

Motion approved unanimously.

- 2. Discussion and possible action regarding the following items:
 - TCERA and TCERA Property, Inc. Preliminary Financial Statements December 31, 2021 a. The Committee reviewed the preliminary financial statements after accruals and before interest posting for December.
 - No action taken.
 - **Ratify Retirement Administrator Expenses** b.

Ms. Warner gave a brief review of why the expenses of both the Retirement Administrator and the Assistant Retirement Administrator must be ratified by the Administrative Committee. Motion to ratify Retirement Administrator expenses as presented.

Motion: Hernandez Second: Polk Motion approved unanimously.

c. Ratify Administrator Action for Surplus Equipment

Ms. Warner informed the committee that the policy regarding the disposal of TCERA assets that grants the Administrator the authority to make decisions regarding equipment with an estimated value of \$500 or less per item. The policy also requires that such actions be reported to the Administrative Committee for ratification.

Motion to ratify Administrator actions regarding disposal of inoperable and obsolete equipment.

Motion: Cook Second: Polk Motion approved unanimously.

d. Actuarial Audit by Roeder Financial

The Committee reviewed the Actuarial Audit report from Roeder Financial. Ms. Malison stated that overall staff is very pleased with the audit. There were no significant findings or problems with the calculations and there were only minor suggestions. The Committee also reviewed the letter from Cheiron and their responses to the audit recommendations made from Roeder Financial. After much discussion, the Committee agreed that Cheiron should focus more on the PEPRA (tier 4) analysis and requested staff to invite Roeder Financial and Cheiron to the earliest Board meeting when both actuaries would be available for further discussion.

V. UPCOMING MEETING

1. Wednesday, March 23, 2022 @ 10:00 a.m.

VI. ADJOURN

The meeting was adjourned at 11:57 a.m.

Approved: _____

David Kehler, Chair

Date



COUNTY OF TULARE BOARD OF RETIREMENT

Leanne Malison Retirement Administrator

136 N AKERS STREET VISALIA, CALIFORNIA 93291 TELEPHONE (559) 713-2900 FAX (559) 730-2631 www.tcera.org

TCERA, Board of Retirement Administrative Committee

Agenda Item # IV.2.a.

Agenda Date: March 23, 2022

Subject: TCERA and TCERA Property, Inc. Financial Statements – December 31, 2021 and January 31, 2022

Requests:

That the Administrative Committee:

- 1. Review the TCERA and TCERA Property, Inc. Financial Statements for December 31, 2021 and January 31, 2022.
- 2. Forward the Financial Reports to the Board of Retirement with a recommendation for approval.

Summary:

The December 31, 2021 and January 31, 2022 TCERA and TCERA Property, Inc. Financial Statements have been prepared by TCERA accounting staff for the Committee's review.

Prepared by: Mary Warner

TCERA and TCERA Property, Inc. Combined Balance Sheet Comparison As of December 31, 2021

 \mathbb{Z}

| | | | Final | | | | | |
|----------|---|------------------|------------------|----------------|----------|------------------|----------------|----------|
| | | Dec 31, 21 | Nov 30, 21 | \$ Change | % Change | Dec 31, 20 | \$ Change | % Change |
| ASSETS | | | | | | | | |
| Curi | rent Assets | | | | | | | |
| | Checking/Savings | | | | | | | |
| (Note 1) | 1110 · Cash in County Treasury | 13,321,076.73 | 19,327,970.29 | -6,006,893.56 | -31.08% | 47,673,469.54 | -34,352,392.81 | -72.06% |
| (Note 2) | 1120 · Cash in Custodial Account | 49,489,035.21 | 25,730,134.35 | 23,758,900.86 | 92.34% | 39,227,605.70 | 10,261,429.51 | 26.16% |
| (Note 3) | 1130 · Short Term Investments | 28,406,889.42 | 35,333,493.01 | -6,926,603.59 | -19.60% | 6,539,752.17 | 21,867,137.25 | 334.37% |
| | 1140 · Securities Lending Collateral | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | 1150 · Impaired Assets | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | Total Checking/Savings | 91,217,001.36 | 80,391,597.65 | 10,825,403.71 | 13.47% | 93,440,827.41 | -2,223,826.05 | -2.38% |
| | Other Current Assets | | | | | | | |
| | 1310 · Fixed Income - Market | 448,463,201.07 | 521,432,213.61 | -72,969,012.54 | -13.99% | 492, 197, 355.56 | -43,734,154.49 | -8.89% |
| | 1340 · Equities - Market | 1,004,343,758.91 | 966,100,453.09 | 38,243,305.82 | 3.96% | 895,427,010.92 | 108,916,747,99 | 12.16% |
| (Note 4) | 1375 · Real Assets | 316,022,503.43 | 251,834,337.96 | 64,188,165.47 | 25.49% | 180,468,196.35 | 135,554,307.08 | 75.11% |
| (Note 5) | 1385 · Hedge Funds | 0,00 | 0.00 | 0.00 | 0.00% | 93,982,33 | -93,982.33 | -100.00% |
| | 1386 · Private Equity | 114,224,080.18 | 116,729,487.25 | -2,505,407.07 | -2.15% | 84,371,791.03 | 29,852,289.15 | 35.38% |
| | 1388 · Private Credit | 107,206,794.00 | 108,603,510.00 | -1,396,716.00 | -1.29% | 84,971,872.00 | 22,234,922.00 | 26.17% |
| | Total Other Current Assets | 1,990,260,337.59 | 1,964,700,001.91 | 25,560,335.68 | 1.30% | 1,737,530,208.19 | 252,730,129.40 | 14.55% |
| Tota | I Current Assets | 2,081,477,338.95 | 2,045,091,599.56 | 36,385,739.39 | 1.78% | 1,830,971,035.60 | 250,506,303.35 | 13.68% |
| Fixe | d Assets | | | | | | | |
| | 1501 · Building and Improvements | 1,178,366.03 | 1,178,366.03 | 0.00 | 0.00% | 1,178,366.03 | 0.00 | 0.00% |
| | 1505 · Office Equipment & Computer Sys | 178,552.49 | 178,552.49 | 0.00 | 0.00% | 153,661.50 | 24,890.99 | 16.20% |
| | 1506 · Project in Process CPAS | 2,755,095.55 | 2,755,095.55 | 0.00 | 0.00% | 2,755,095.55 | 0.00 | 0.00% |
| | 1511 · Accumulated Depreciation | -3,109,385,42 | -3,109,385.42 | 0.00 | 0.00% | -2,786,542.42 | -322,843.00 | -11.59% |
| | 1512 · Land | 370,345.69 | 370,345.69 | 0.00 | 0.00% | 370,345.69 | 0.00 | 0.00% |
| Tota | I Fixed Assets | 1,372,974.34 | 1,372,974.34 | 0.00 | 0.00% | 1,670,926.35 | -297,952.01 | -17.83% |
| Oth | er Assets | | | | | | | |
| (Note 6) | 1710 · Open Trades Sales | 26,669,950.99 | 15,227,495.84 | 11,442,455 15 | 75.14% | 12,806,793.00 | 13,863,157.99 | 108.25% |
| | 1730 · Investment Income Receivable | 833,339.14 | 1,078,070.93 | -244,731.79 | -22.70% | 848,674.02 | -15,334.88 | -1.81% |
| | 1735 · Real Assets Income Receivable | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| (Note 7) | 1750 · Members Contribution Receivable | 981,148.43 | 0.00 | 981,148.43 | 100.00% | 838,866.41 | 142,282.02 | 16.96% |
| (Note 7) | 1770 · Employer Contribution Receivable | 119,105.91 | 0.00 | 119,105.91 | 100.00% | 222,101.12 | -102,995.21 | -46.37% |
| | 1780 · Advances Rec- Holding Corp | 428,230.00 | 437,230.00 | -9,000.00 | -2.06% | 536,230.00 | -108,000.00 | -20.14% |
| | 1785 · Pension Deaths Receivables | 0.00 | 0.00 | 0.00 | 0.00% | 109.93 | -109.93 | ~100.00% |
| | 1790 · Other Receivables | 5,265.00 | 5,556.01 | -291.01 | -5.24% | 8,739.99 | -3,474.99 | -39.76% |
| Tota | al Other Assets | 29,037,039.47 | 16,748,352.78 | 12,288,686.69 | 73.37% | 15,261,514.47 | 13,775,525.00 | 90.26% |
| TOTAL A | SSETS | 2,111,887,352.76 | 2,063,212,926.68 | 48,674,426.08 | 2.36% | 1,847,903,476.42 | 263,983,876.34 | 14.29% |

TCERA and TCERA Property, Inc. Combined Balance Sheet Comparison As of December 31, 2021

| | | | Final | | | | | |
|----------|--|------------------|------------------|----------------|----------|------------------|-----------------|----------|
| | | Dec 31, 21 | Nov 30, 21 | \$ Change | % Change | Dec 31, 20 | \$ Change | % Change |
| LIABILIT | ES & EQUITY | 1/81 | | | | | | |
| Liat | ilities | | | | | | | |
| | Current Liabilities | | | | | | | |
| | 2010 · Sec Lending Collateral Paya | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| (Note 8) | 2020 · Open Trades - Purchases | 67,572,411.94 | 49,419,032.47 | 18,153,379.47 | 36.73% | 17,645,161.06 | 49,927,250.88 | 282.95% |
| (Note 7) | 2030 · Accounts Payable - Inv | 693,480.67 | 0.00 | 693,480.67 | 100.00% | 729,928.27 | -36,447.60 | -4.99% |
| (Note 7) | 2040 · Refunds Payable | 5,718,404.40 | 4,098,875.89 | 1,619,528.51 | 39.51% | 3,706,735.89 | 2,011,668.51 | 54.27% |
| (Note 7) | 2050 · Other Payables | 253,397.07 | 0.00 | 253,397.07 | 100.00% | 149,204.29 | 104,192.78 | 69.83% |
| (Note 7) | 2100 · Payroll Liabilities | 47,630.15 | 0.00 | 47,630.15 | 100.00% | 41,746.62 | 5,883.53 | 14.09% |
| | Total Other Current Liabilities | 74,285,324.23 | 53,517,908.36 | 20,767,415.87 | 38.81% | 22,272,776.13 | 52,012,548.10 | 233.53% |
| | Total Current Liabilities | 74,285,324.23 | 53,517,908.36 | 20,767,415.87 | 38.81% | 22,272,776.13 | 52,012,548.10 | 233.53% |
| | Long Torm Linkilliting | | | | | | | |
| | Long Term Liabilities | 445 000 00 | 445 000 00 | 0.00 | 0.000/ | | | |
| | 2060 · Accrual-Benefits at Termination | 115,302.68 | 115,302.68 | 0.00 | 0.00% | 103,551.79 | 11,750.89 | 11.35% |
| | 2070 · Advances Payable - TCERA | 428,230.00 | 437,230.00 | -9,000.00 | -2.06% | 536,230.00 | -108,000.00 | -20.14% |
| _ | Total Long Term Liabilities | 543,532.68 | 552,532.68 | -9,000.00 | -1.63% | 639,781.79 | -96,249.11 | -15.04% |
| Tota | al Liabilities | 74,828,856.91 | 54,070,441.04 | 20,758,415.87 | 38.39% | 22,912,557.92 | 51,916,298.99 | 226.58% |
| Equ | ity | | | | | | | |
| | 3110 · Member Deposit Reserve | 351,736,907.14 | 331,389,984.54 | 20,346,922.60 | 6.14% | 329,911,156.12 | 21,825,751.02 | 6.62% |
| (Note 9) | 3120 · Other Reserves - Unapportioned | -10,553,333.56 | 2,550,693.48 | -13,104,027.04 | -513.74% | 3,127,636.01 | -13,680,969.57 | -437.42% |
| | 3210 · Employer Advance Reserves | 966,736,747.33 | 893,879,609.33 | 72,857,138.00 | 8.15% | 902,630,252.28 | 64,106,495.05 | 7.10% |
| | 3310 · Retiree Reserves | 446,249,372.92 | 469,528,895.11 | -23,279,522.19 | -4.96% | 433,025,431.67 | 13,223,941.25 | 3.05% |
| | 3320 · Supp. Retiree Benefit Reserve | 113,579,360.95 | 108,446,851.09 | 5,132,509.86 | 4.73% | 107,567,108.49 | 6,012,252.46 | 5.59% |
| | 3410 · Contingency Reserve | 64,035,077.22 | 60,735,602.01 | 3,299,475.21 | 5.43% | 55,437,104.30 | 8,597,972.92 | 15.51% |
| | 3510 · Market Stabilization | 104,263,467.00 | 108,713,808.00 | -4,450,341.00 | 4.09% | -7,569,506.00 | 111,832,973.00 | 1477.41% |
| | 3810 · Income Summary Account | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | 3900 · Retained Earnings | -59,862,746.31 | 939,409.13 | -60,802,155.44 | 6472.38% | -204,309,472.39 | 144,446,726.08 | 70.70% |
| | Net Income | 60,873,643.16 | 32,957,632.95 | 27,916,010.21 | 84.70% | 205,171,208.02 | -144,297,564.86 | -70.33% |
| Tota | al Equity | 2,037,058,495.85 | 2,009,142,485.64 | 27,916,010.21 | 1.39% | 1,824,990,918.50 | 212,067,577.35 | 11.62% |
| TOTAL L | IABILITIES & EQUITY | 2,111,887,352.76 | 2,063,212,926.68 | 48,674,426.08 | 2.36% | 1,847,903,476.42 | 263,983,876.34 | 14.29% |

Notes:

Note 1 Prior year transfer to 421 Fund \$20M from Cash Reserve at Custodial Bank

Note 2 Rebalancing Assets and cash flow needs

Note 3 Timing of managers' short term positions

Note 4 Fund SSGA US REIT \$40M and capital calls from Invesco and ARA

Note 5 Hedge Funds - accounts closed

Note 6 Increase in Open Trade Sales over last mo. and last yr.

Note 7 Accruals recorded for 12/31

Note 8 Increase in Open Trade Purchases over last mo. and last yr.

Note 9 Interim interest payments/adjustments and private market adjustments prior to period close

TCERA and TCERA Property, Inc. Combined Comparative Profit and Loss December 31, 2021

| IINCI | ~ | 2 |
|-------|---|---|
| Final | | |

| | - | Final | | | | | | | | | |
|----------|---------------------------------------|---------------|---------------|--------------|-----------|----------------|----------------|---------------|----------|--|--|
| | | Dec 21 | Nov 21 | \$ Change | % Change | Jul - Dec 21 | Jul - Dec 20 | \$ Change | % Change | | |
| Ordin | ary Income/Expense | | | | 0 | | | | | | |
| 1 | ncome | | | | | | | | | | |
| | 4110 · Interest Income | 238,972.32 | 319,428.38 | -80,456.06 | -25.19% | 1,638,693.24 | 1,998,098.81 | -359,405.57 | -17.99% | | |
| | 4120 · Dividend Income | 344,312.43 | 415,770.52 | -71,458.09 | -17.19% | 1,809,691.33 | 2,013,766.65 | -204,075.32 | -10.13% | | |
| | 4130 · Real Assets Income | 0.00 | 0.00 | 0.00 | 0.0% | 1,846,891.34 | 1,434,936.61 | 411,954.73 | 28.71% | | |
| (Note 1) | 4140 · Other Investment Income | 3,274,849.50 | 202,534.18 | 3,072,315.32 | 1,516.94% | 4,094,458.87 | 5,461,712.17 | -1,367,253.30 | -25.03% | | |
| | 4200 · Lease Payments from TCERA | 15,640.00 | 15,640.00 | 0.00 | 0.0% | 93,840.00 | 93,840.00 | 0.00 | 0.0% | | |
| | 4310 · Commission Rebates | 241.06 | 0.00 | 241.06 | 100.0% | 458.88 | 1,385.81 | -926.93 | -66.89% | | |
| (Note 2) | 4410 · Securities Lending Income | 7,772.55 | 10,004.16 | -2,231.61 | -22.31% | 44,528.56 | 25,403.77 | 19,124.79 | 75.28% | | |
| (Note 3) | 4510 · Realized Gains/Losses | 8,942,718.27 | 6,016,288.94 | 2,926,429.33 | 48.64% | 75,417,103.52 | 42,976,900.83 | 32,440,202.69 | 75.48% | | |
| | 4530 · Gn/Ls Disposal of Fixed Asset | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% | | |
| (Note 4) | 4610 · Employee Contributions | 2,992,457.82 | 2,852,182.21 | 140,275.61 | 4.92% | 13,120,494.85 | 11,999,472.66 | 1,121,022.19 | 9.34% | | |
| (Note 4) | 4620 · Employer Contributions | 463,707.77 | 233,569.96 | 230,137.81 | 98.53% | 40,812,441.64 | 37,275,599.77 | 3,536,841.87 | 9.49% | | |
| T I | Total Income | 16,280,671.72 | 10,065,418.35 | 6,215,253.37 | 61.75% | 138,878,602.23 | 103,281,117.08 | 35,597,485.15 | 34.47% | | |
| | | | | | | | | | | | |
| | | | | | | | | | | | |
| F | Expense | | | | | | | | | | |
| | 5110 · Benefit Payments | 7,967,616.71 | 8,029,639.68 | -62,022.97 | -0.77% | 48,047,374.75 | 45,621,978.02 | 2,425,396.73 | 5.32% | | |
| (Note 5) | 5120 · Refunds | 1,908,072.62 | 313,533.66 | 1,594,538.96 | 508.57% | 3,713,813.21 | 1,469,609.58 | 2,244,203.63 | 152.71% | | |
| | 5130 · Death Retiree ROC | 0.00 | 0.00 | 0.00 | 0.0% | 13,508.23 | 76,646.62 | -63,138.39 | -82.38% | | |
| | 5140 · SDA Payments | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% | | |
| (Note 6) | 5210 · Investment Management Fees | 1,633,031.52 | 382,414.82 | 1,250,616.70 | 327.03% | 3,354,808.10 | 2,988,824.93 | 365,983.17 | 12.25% | | |
| (Note 6) | 5250 · Inv. Consultant/Custodial Fees | 202,558.32 | 22,500.00 | 180,058.32 | 800.26% | 290,878,40 | 277,279.31 | 13,599.09 | 4.9% | | |
| | 5270 · Securities Lending Expense | 716.87 | 716.66 | 0.21 | 0.03% | 4,036.43 | 3,972,35 | 64.08 | 1.61% | | |
| | 5275 · Real Assets Investment Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% | | |
| (Note 6) | 5276 · Real Assets Mgr Fees | 609,574.12 | 0.00 | 609,574.12 | 100.0% | 1,098,584.31 | 842,733.31 | 255,851.00 | 30.36% | | |
| (Note 7) | 5280 · Other Investment Expense | 150,280.69 | 7,638.83 | 142,641.86 | 1,867.33% | 282,241.20 | 131,292.38 | 150,948.82 | 114.97% | | |
| | 5410 · Actuarial Study Fees | 50,838.75 | 52,219.14 | -1,380.39 | -2.64% | 104,027.89 | 113,866.50 | -9,838.61 | -8.64% | | |
| | 5450 · Compensated Benefit Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% | | |
| | 5500 · Administrative Expense | 233,465.79 | 228,325.74 | 5,140.05 | 2.25% | 1,134,344.57 | 1,106,221.14 | 28,123.43 | 2.54% | | |
| | 5750 · TCERA Property Admin Expense | 3,147,14 | 1,590.58 | 1,556.56 | 97.86% | 24,270.04 | 22,228.93 | 2,041.11 | 9.18% | | |
| | 5910 · Depreciation of Fixed Assets | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% | | |
| | Total Expense | 12,759,302.53 | 9.038.579.11 | 3,720,723,42 | 41.17% | 58,067,887.13 | 52,654,653.07 | 5 442 224 00 | 10.089/ | | |
| | i our importor | 12,100,002.00 | 0,000,070.11 | 0,120,120.42 | 41.1770 | 30,007,007.13 | 52,004,000.07 | 5,413,234.06 | 10.28% | | |
| Net O | ordinary Income | 3,521,369.19 | 1,026,839.24 | 2,494,529.95 | -242.93% | 80,810,715.10 | 50,626,464.01 | 30,184,251.09 | 59.62% | | |

TCERA and TCERA Property, Inc. Combined Comparative Profit and Loss December 31, 2021

| | | | Final | | | | | |
|--|---------------|----------------|---------------|----------|----------------|----------------|-----------------|----------|
| | Dec 21 | Nov 21 | \$ Change | % Change | Jul - Dec 21 | Jul - Dec 20 | \$ Change | % Change |
| Other Income/Expense | | | | | | | | , |
| Other Income | | | | | | | | |
| 4520 · Unrealized Gains/Losses | 24,394,641.02 | -26,397,737.66 | 50,792,378.68 | 192.41% | -19,937,071.94 | 154,544,744.01 | -174,481,815.95 | -112.9% |
| 4525 · Unrealized Gns/Ls Building/Land | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| 5000 · Other Income | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| Total Other Income | 24,394,641.02 | -26,397,737.66 | 50,792,378.68 | 192.41% | -19,937,071.94 | 154,544,744.01 | -174,481,815.95 | -112.9% |
| Other Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| Net Other Income | 24,394,641.02 | -26,397,737.66 | 50,792,378.68 | 192.41% | -19,937,071.94 | 154,544,744.01 | -174,481,815.95 | -112.9% |
| Net Income | 27,916,010.21 | -25,370,898.42 | 53,286,908.63 | 210.03% | 60,873,643.16 | 205,171,208.02 | -144,297,564.86 | -70.33% |

Notes:

Note 1 Other Investment Income increase over last mo. in Priv. Credit

Note 2 Securities Lending Income increase over last year

Note 3 Realized Gains increase over last mo. and last year in Public Equity

Note 4 Accruals for EE & ER Contributions

Note 5 Semi-Annual Refund Accrual

Note 6 Accruals recorded for 12/31

Note 7 Increase in Other Investment Expenses over last mo. and last year in Priv. Credit

* See Budget report for detail of Administrative expenses (5500 and 5750)

TCERA and TCERA Property, Inc. BUDGET VS ACTUAL December 2021 Accounting Period 6

50.00% of the Current Fiscal Year Budget

Final

| | | Dec 21 | Jul - Dec 20 | Jul - Dec 21 | Annual Budget | Remaining Budget | % Annual Budget |
|-----------|--|------------|--------------|--------------|---------------|------------------|-----------------|
| 5500 · Ad | ministrative Expense | | | | | | N |
| | 5505 · (6001) Allocated Salaries | 112,570.89 | 404,598.08 | 438,142.82 | 1,025,313.00 | 587,170.18 | 42.73% |
| | 5510 · (6002) Overtime | 0.00 | 0.00 | 0.00 | 1,000.00 | 1,000,00 | 0.00% |
| (Note 1) | 5515 · (6003) Other Pay | -13,661.70 | 4,334.03 | 7,518.48 | 28,672.00 | 21,153.52 | 26.22% |
| | 5520 · (6004) Benefits | 15,548.45 | 68,436.31 | 67,111.86 | 159,376.00 | 92,264.14 | 42.11% |
| (Note 2) | 5525 · (6005) Extra-Help | 2,913.74 | 0.00 | 7,461.59 | 500.00 | -6,961.59 | 1492.32% |
| | 5530 · (6006) Sick Leave Buy Back | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 | 0.00% |
| | 5535 · (6011) Retirement- Co. Port. | 12,029.10 | 51,817.94 | 56,424.69 | 136,235.00 | 79,810.31 | 41.42% |
| | 5540 · (6012) Social Security | 6,744.17 | 29,154.75 | 31,149.37 | 76,404.00 | 45,254.63 | 40.77% |
| | 5545 · (1024) POB Cost | 9,021.73 | 7,071.24 | 35,087.44 | 87,249.00 | 52,161.56 | 40.22% |
| (Note 3) | 5550 · (6008) Board Fees-Per Diem Pmts | 0.00 | 9,300.00 | 10,119.10 | 20,000.00 | 9,880.90 | 50.60% |
| | 5551 · Communications | | | | | | |
| | 5552 · (7005) Communications | 520.69 | 2,775.79 | 2,567.69 | 8,700.00 | 6,132.31 | 29.51% |
| | 5640 · (7005) Co. Telecommunication | 0.00 | 4,059.57 | 2,832.12 | 12,039.00 | 9,206.88 | 23.52% |
| | Total 5551 · Communications | 520.69 | 6,835.36 | 5,399.81 | 20,739.00 | 15,339.19 | 26.04% |
| | 5553 · Data Processing | | | | | | |
| | 5650 · (7044) ICT Qtrly - Data Process | 7,816.49 | 21,018.16 | 28,436.82 | 104,821.00 | 76,384.18 | 27.13% |
| | 5697 · (7044) Computer Exp Hdwr/Sftwr | 0.00 | 10,506.54 | 6,614.71 | 32,138.00 | 25,523.29 | 20.58% |
| | 5698 · (7044) WSI/Q2 Digital | 0.00 | 1,750.00 | 0.00 | 3,500.00 | 3,500.00 | 0.00% |
| | Total 5553 · Data Processing | 7,816.49 | 33,274.70 | 35,051.53 | 140,459.00 | 105,407.47 | 24.95% |
| | 5555 · (7009) Household Expense | 1,321,21 | 7,101.88 | 8,103.81 | 17,380.00 | 9,276.19 | 46.63% |
| (Note 4) | 5560 · (7010) Insurance | 0.00 | 54,402.00 | 58,624.00 | 59,400.00 | 776.00 | 98.69% |
| | 5565 · (7011) Unemployment Insurance | 0.00 | 0.00 | 14.67 | 6,000.00 | 5,985.33 | 0.00% |
| | 5570 · (7021) Maintenance-Equipment | 96.59 | 666.03 | 645.36 | 3,000.00 | 2,354.64 | 21.51% |
| (Note 5) | 5575 · (7027) Memberships | 0.00 | 6,010.00 | 4,550.00 | 7,100.00 | 2,550.00 | 64.08% |
| | 5580 · (7030) Due Diligence Expense | 0.00 | 0.00 | 2,521,92 | 15,000.00 | 12,478.08 | 16.81% |
| (Note 6) | 5585 · (7036) Office Expense | -55.93 | 6,453.90 | 22,922.29 | 50,080.00 | 27,157.71 | 45.77% |
| | 5586 · (7040) Courier | 118.16 | 556.77 | 472.14 | 1,460.00 | 987.86 | 32.34% |
| | 5590 · Prof & Specialized Exp | | | | | | |
| | 5591 · (7043) Prof & Special Gen Exp | 0.00 | 0.00 | 0.00 | 500.00 | 500.00 | 0.00% |
| (Note 7) | 5592 · (7043) Prof & Spec - Audit | 15,621.72 | 44,570.00 | 41,871.22 | 47,600.00 | 5,728.78 | 87.96% |
| | 5593 · (7043) Prof Sr -Outside Counsel | 9,055.80 | 11,612.70 | 9,055.80 | 75,000.00 | 65,944.20 | 12.07% |
| | Total 5590 · Prof & Specialized Exp | 24,677.52 | 56,182.70 | 50,927.02 | 123,100.00 | 72,172.98 | 41.37% |
| | 5594 · County Counsel Charges | | | | | | |
| | 5661 · (7046) Co Counsel - General Exp | 804.60 | 6,040.10 | 4,559.80 | 40,000.00 | 35,440.20 | 11.40% |
| | 5662 · (7046) Co Counsel - Disability | 3,004.60 | 30,349.00 | 20,056.20 | 93,000.00 | 72,943.80 | 21.57% |
| | Total 5594 · County Counsel Charges | 3,809.20 | 36,389.10 | 24,616.00 | 133,000.00 | 108,384.00 | 18.51% |
| | 5595 · (7049) Prof Exp-Disabilities | 13,187.50 | 29,969.44 | 25,578.08 | 164,000.00 | 138,421.92 | 15.60% |
| | 5600 · (7059) Publications | 0.00 | 4,137.60 | 4,375.36 | 22,000.00 | 17,624.64 | 19.89% |
| | 5675 · (7059) Co. Print Services | 76.37 | 2,444.91 | 541.70 | 14,500.00 | 13,958.30 | 3.74% |
| | Total 5600 · (7059) Publications | 76.37 | 6,582.51 | 4,917.06 | 36,500.00 | 31,582.94 | 13.47% |

TCERA and TCERA Property, Inc. BUDGET VS ACTUAL December 2021 Accounting Period 6 50.00% of the Current Fiscal Year Budget

Final

| | | Dec 21 | Jul - Dec 20 | Jul - Dec 21 | Annual Budget | Remaining Budget | % Annual Budget |
|------------|--|------------|--------------|--------------|---------------|------------------|-----------------|
| | 5605 · (7062) Rent & Lease -Building | 15,640.00 | 93,840.00 | 93,840.00 | 187,680.00 | 93,840.00 | 50.00% |
| | 5610 · (7066) Spec Dept Exp - RIS | 16,245.17 | 107,441.31 | 60,734.30 | 318,120,00 | 257,385.70 | 19.09% |
| | 5615 · (7073) Training | 0.00 | 1,870.00 | 1,830.00 | 15,500.00 | 13,670.00 | 11.81% |
| | 5620 · (7074) Transportation & Travel | 261.98 | 975.56 | 2,294.27 | 32,000.00 | 29,705.73 | 7.17% |
| (Note 8) | 5625 · (7081) Utilities | 1,152.71 | 9,756.34 | 11,026.65 | 20,400.00 | 9,373.35 | 54.05% |
| | 5627 · (7116) Postage - Co. Mail | 3,431.75 | 13,296.61 | 14,379.67 | 46,900.00 | 32,520.33 | 30,66% |
| (Note 4) | 5630 · (7128) Co. Workers Comp Insurance | 0.00 | 59,381.00 | 52,514.00 | 53,500.00 | 986.00 | 98.16% |
| | 5695 · Co. Admin. Services | | | | | | |
| | 5666 · (7719) HR/Risk Services | 0.00 | 0.00 | 0.00 | 11,860.00 | 11,860.00 | 0.00% |
| | 5699 · (7719) Auditors Services | 0.00 | 523.58 | 366.64 | 7,500.00 | 7,133.36 | 4.89% |
| | Total 5695 · Co. Admin. Services | 0.00 | 523.58 | 366.64 | 19,360.00 | 18,993.36 | 1.89% |
| | 5720 - (7421) Interest Expense | 0.00 | 0.00 | 0.00 | 1.00 | 1.00 | 0.00% |
| | 5450 - Compensated Benefit Expense | 0.00 | 0.00 | 0.00 | 26,353.00 | 26,353.00 | 0.00% |
| | 5911 · Depreciation - TCERA | 0.00 | 0.00 | 0.00 | 20,000.00 | 20,000.00 | 0.00% |
| | 5913 · Amortization - CPAS | 0.00 | 0.00 | 0.00 | 325,000.00 | 325,000.00 | 0.00% |
| Total TCEF | RA Administrative Expense | 233,465.79 | 1,106,221.14 | 1,134,344.57 | 3,376,781.00 | 2,242,436.43 | 33.59% |
| 5750 · TCE | RA Property Administrative Expense | | | | | | |
| | 5755 · Fees and Taxes | 0.00 | 25.00 | 25.00 | 200.00 | 175.00 | 12,50% |
| (Note 4) | 5760 · Insurance | 0.00 | 7,452.00 | 7,909.00 | 9,000.00 | 1,091.00 | 87.88% |
| | 5765 · Professional & Spec Services | 0.00 | 0.00 | 0.00 | 2,000.00 | 2,000.00 | 0.00% |
| | 5780 · Courtyards Property Assn Dues | 1,707.67 | 3,415.34 | 3,415.34 | 7,600.00 | 4,184.66 | 44.94% |
| | 5785 · Landscape Service | 655.00 | 2,047.25 | 2,205.00 | 7,800_00 | 5,595.00 | 28.27% |
| | 5790 · Security Monitoring | 428,43 | 808.58 | 1,430.03 | 7,750.00 | 6,319.97 | 18.45% |
| | 5795 · Maintenance & Improve - Bldg | 0.00 | 6,105.77 | 6,418.00 | 244,365.00 | 237,947.00 | 2.63% |
| | 5797 · Utilities | 356.04 | 2,374.99 | 2,867.67 | 7,000.00 | 4,132,33 | 40.97% |
| | 5912 · Depreciation - TCERA Property | 0.00 | 0.00 | 0.00 | 70,400.00 | 70,400.00 | 0.00% |
| Total TCER | RA Property, Inc. Administrative Expense | 3,147.14 | 22,228.93 | 24,270.04 | 356,115.00 | 331,844.96 | 6.82% |
| Total TCEF | RA & TCERA Property Administrative Expense | 236,612.93 | 1,128,450.07 | 1,158,614.61 | 3,732,896.00 | 2,574,281.39 | 31.04% |

Notes:

| Note 1 | County reimbursement of \$15k for one time EE payments |
|--------|--|
| Note 2 | Extra Help Office Assistant hired until full time replacement is found |
| Note 3 | Quarterly Per Diem paid |
| Note 4 | Annual Insurance premiums paid |
| Note 5 | Annual memberships paid: SACRS, NCPERS |
| Note 6 | Reimbursement of Tax Penalty paid in Nov. |
| Note 7 | Annual audit expense paid in the first half of fiscal year |
| Note 8 | Increased electricity costs during summer months |
| | |

| Accrued Actuarial Liability | 1,957,985,000 | | | | |
|-----------------------------|----------------|--|--|--|--|
| June 30, 2021 | | | | | |
| 21% of AAL | 4,111,768.50 | | | | |
| 50.00% | 2,055,884.25 | | | | |
| Expenses to date | (1,158,614.61) | | | | |
| (Over)/Under | 897,269.64 | | | | |

TCERA and TCERA Property, Inc. Combined Balance Sheet Comparison As of January 31, 2022

| | | Jan 31, 22 | Dec 31, 21 | \$ Change | % Change | Jan 31, 21 | \$ Change | % Change |
|----------|--|-----------------------|------------------|----------------|----------|------------------|----------------|-------------|
| ACCETO | | | | | | | - + enange | ,, oriningo |
| ASSETS | | | | | | | | |
| Curr | ent Assets | | | | | | | |
| (Nete 1) | Checking/Savings | 0.057.000.07 | 40.004.070.70 | 0 000 754 00 | 50.000/ | | | |
| (Note 1) | 1110 · Cash in County Treasury 1120 · Cash in Custodial Account | 6,357,322.37 | 13,321,076.73 | -6,963,754.36 | -52.28% | 41,598,863.48 | -35,241,541.11 | -84.72% |
| (Note 2) | 1130 · Short Term investments | 51,931,407.29 | 49,489,035.21 | 2,442,372.08 | 4.94% | 42,585,904.86 | 9,345,502.43 | 21.95% |
| (Note 2) | 1140 · Securities Lending Collateral | 26,881,202.37 0.00 | 28,406,889.42 | -1,525,687.05 | -5.37% | 10,319,493.09 | 16,561,709.28 | 160.49% |
| | 1150 · Impaired Assets | 0.00 | 0.00 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | • | | | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | Total Checking/Savings | 85,169,932.03 | 91,217,001.36 | -6,047,069.33 | -6.63% | 94,504,261.43 | -9,334,329.40 | -9.88% |
| | Other Current Assets | | | | | | | |
| | 1310 · Fixed Income - Market | 426,630,416.05 | 448,463,201.07 | -21,832,785.02 | -4.87% | 487,452,958.63 | -60,822,542.58 | -12.48% |
| | 1340 · Equities - Market | 943,326,280.41 | 1,004,343,758.91 | -61,017,478.50 | -6.08% | 896,809,902.61 | 46,516,377.80 | 5.19% |
| (Note 3) | 1375 · Real Assets | 344,377,870.76 | 316,022,503.43 | 28,355,367.33 | 8.97% | 180,503,855.21 | 163,874,015.55 | 90.79% |
| (Note 4) | 1385 · Hedge Funds | 0.00 | 0.00 | 0.00 | 0.00% | 93,982,33 | -93,982.33 | -100.00% |
| | 1386 · Private Equity | 119,875,036.90 | 114,224,080.18 | 5,650,956.72 | 4.95% | 84,748,788.80 | 35,126,248.10 | 41.45% |
| | 1388 · Private Credit | 108,847,922.00 | 107,206,794.00 | 1,641,128.00 | 1.53% | 84,971,872.00 | 23,876,050.00 | 28.10% |
| | Total Other Current Assets | 1,943,057,526.12 | 1,990,260,337.59 | -47,202,811.47 | -2.37% | 1,734,581,359.58 | 208,476,166.54 | 12.02% |
| Tota | Current Assets | 2,028,227,458.15 | 2,081,477,338.95 | -53,249,880.80 | -2.56% | 1,829,085,621.01 | 199,141,837.14 | 10.89% |
| Fixe | d Assets | | | | | | | |
| | 1501 · Building and Improvements | 1,178,366.03 | 1,178,366.03 | 0.00 | 0.00% | 1,178,366.03 | 0.00 | 0.00% |
| | 1505 · Office Equipment & Computer Sys | 178,552.49 | 178,552.49 | 0.00 | 0.00% | 153,661.50 | 24,890.99 | 16.20% |
| | 1506 · Project in Process CPAS | 2,755,095.55 | 2,755,095.55 | 0,00 | 0.00% | 2,755,095.55 | 0.00 | 0.00% |
| | 1511 · Accumulated Depreciation | -3,109,385.42 | -3,109,385.42 | 0.00 | 0.00% | -2,786,542,42 | -322,843.00 | -11.59% |
| | 1512 · Land | 370,345.69 | 370,345.69 | 0.00 | 0.00% | 370,345.69 | 0.00 | 0.00% |
| Tota | Fixed Assets | 1,372,974.34 | 1,372,974.34 | 0.00 | 0.00% | 1,670,926.35 | -297,952.01 | -17.83% |
| Othe | er Assets | | | | | | | |
| (Note 5) | 1710 · Open Trades Sales | 22,745,133.97 | 26,669,950.99 | -3,924,817.02 | -14.72% | 17,537,867.05 | 5,207,266.92 | 29.69% |
| | 1730 · Investment Income Receivable | 823,420.35 | 833,339.14 | -9,918.79 | -1.19% | 835,646.77 | -12,226.42 | -1.46% |
| | 1735 · Real Assets Income Receivable | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| (Note 6) | 1750 · Members Contribution Receivable | 0.00 | 981,148.43 | -981,148.43 | -100.00% | 0.00 | 0.00 | 0.00% |
| (Note 6) | 1770 · Employer Contribution Receivable | 0.00 | 119,105.91 | -119,105.91 | -100.00% | 0.00 | 0.00 | 0.00% |
| | 1780 · Advances Rec- Holding Corp | 419,230.00 | 428,230.00 | -9,000.00 | -2.10% | 527,230.00 | -108,000.00 | -20.48% |
| | 1785 · Pension Deaths Receivables | 0.00 | 0.00 | 0.00 | 0.00% | 836.93 | -836.93 | -100.00% |
| (Note 7) | 1790 · Other Receivables | -279.76 | 5,265.00 | -5,544.76 | -105.31% | 8,454.69 | -8,734.45 | -103.31% |
| Tota | I Other Assets | 23,987,504.56 | 29,037,039.47 | -5,049,534.91 | -17.39% | 18,910,035.44 | 5,077,469.12 | 26.85% |
| TOTAL A | SSETS | 2,053,587,937.05 | 2,111,887,352.76 | -58,299,415.71 | -2.76% | 1,849,666,582.80 | 203,921,354.25 | 11.03% |

TCERA and TCERA Property, Inc. Combined Balance Sheet Comparison As of January 31, 2022

| | - | Jan 31, 22 | Dec 31, 21 | \$ Change | % Change | Jan 31, 21 | \$ Change | % Change |
|----------------------------|---|------------------|------------------|----------------|-----------------|--------------------------|--------------------------|----------|
| LIABILITI | ES & EQUITY | 2 A | | | | | | |
| Liab | ilities | | | | | | | |
| | Current Liabilities | | | | | | | |
| | 2010 · Sec Lending Collateral Paya | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| (Note 8) | 2020 · Open Trades - Purchases | 48,865,531.32 | 67,572,411.94 | -18,706,880.62 | -27.68% | 24,259,342.04 | 24,606,189.28 | 101.43% |
| (Note 6) | 2030 · Accounts Payable - Inv | 421,665.43 | 693,480.67 | -271,815.24 | -39.20% | 547,203.65 | -125,538.22 | -22.94% |
| (Note 6) | 2040 · Refunds Payable | 5,718,404.40 | 5,718,404.40 | 0.00 | 0.00% | 3,706,735.89 | 2,011,668.51 | 54.27% |
| (Note 6) | 2050 · Other Payables | 0.00 | 253,397.07 | -253,397.07 | -100.00% | -389.82 | 389.82 | 100.00% |
| (Note 6) | 2100 · Payroll Liabilities | 0.00 | 47,630.15 | -47,630.15 | -100.00% | 0.00 | 0.00 | 0.00% |
| | Total Other Current Liabilities | 55,005,601.15 | 74,285,324.23 | -19,279,723.08 | -25.95% | 28,512,891.76 | 26,492,709.39 | 92.92% |
| | Total Current Liabilities | 55,005,601.15 | 74,285,324.23 | -19,279,723.08 | -25.95% | 28,512,891.76 | 26,492,709.39 | 92.92% |
| | | | | | | | | |
| | Long Term Liabilities 2060 · Accrual-Benefits at Termination | 115.302.68 | 115.302.68 | 0.00 | 0.000/ | 102 551 70 | 44 750 00 | 11.35% |
| | 2000 · Accrual-Benefits at Termination 2070 · Advances Payable - TCERA | 419,230.00 | 428,230.00 | -9,000.00 | 0.00% -2.10% | 103,551.79 527,230.00 | 11,750.89 -108,000.00 | -20.48% |
| | Total Long Term Liabilities | 534,532.68 | 543,532.68 | -9,000.00 | -2.10% | 630,781,79 | -96,249,11 | -15.26% |
| T -4 | - | | | | | | | ÷ |
| lota | al Liabilities | 55,540,133.83 | 74,828,856.91 | -19,288,723.08 | -25.78% | 29,143,673,55 | 26,396,460.28 | 90.57% |
| Equ | ity | | | | | | | |
| | 3110 · Member Deposit Reserve | 349,353,792.38 | 351,736,907.14 | -2,383,114.76 | -0.68% | 328,431,768.25 | 20,922,024.13 | 6.37% |
| (Note 9) | 3120 · Other Reserves - Unapportioned | -10,576,288.49 | -10,553,333.56 | -22,954.93 | 0.22% | 3,127,636.01 | -13,703,924.50 | -438_16% |
| | 3210 · Employer Advance Reserves | 960,953,327.47 | 966,736,747.33 | -5,783,419.86 | -0.60% | 899,363,956.29 | 61,589,371.18 | 6.85% |
| | 3310 · Retiree Reserves | 454,438,862.47 | 446,249,372.92 | 8,189,489.55 | 1.84% | 437,771,115.53 | 16,667,746.94 | 3.81% |
| | 3320 · Supp. Retiree Benefit Reserve | 113,579,360.95 | 113,579,360.95 | 0.00 | 0.00% | 107,567,108.49 | 6,012,252.46 | 5.59% |
| | 3410 · Contingency Reserve | 64,035,077.22 | 64,035,077.22 | 0.00 | 0.00% | 55,437,104.30 | 8,597,972.92 | 15.51% |
| | 3510 · Market Stabilization | 104,263,467.00 | 104,263,467.00 | 0.00 | 0.00% | -7,569,506.00 | 111,832,973.00 | 1477.41% |
| | 3810 · Income Summary Account | 0.00 | 0.00 | 0.00 | 0.00% | 0.00 | 0.00 | 0.00% |
| | 3900 · Retained Earnings | -59,862,746.31 | -59,862,746.31 | 0.00 | 0.00% | -204,309,472.39 | 144,446,726.08 | 70.70% |
| | Net Income | 21,862,950.53 | 60,873,643.16 | -39,010,692.63 | -64.09% | 200,703,198.77 | -178,840,248.24 | -89.11% |
| Tot | al Equity | 1,998,047,803.22 | 2,037,058,495.85 | -39,010,692.63 | -1.92% | 1 820,522,909.25 | 177,524,893.97 | 9.75% |
| TOTAL LIABILITIES & EQUITY | | 2,053,587,937.05 | 2,111,887,352.76 | -58,299,415.71 | -2.76% | 1,849,666,582.80 | 203,921,354.25 | 11.03% |

Notes:

Note 1 The transfer to 421 Fund from Cash Reserve at Custodial Bank occurs in Feb this yr.

Note 2 Timing of managers' short term positions

Note 3 Increase over last year due to 3 new Real Asset managers: SSGA US REIT, Invesco and ARA

Note 4 Hedge Funds - accounts closed

Note 5 Decrease in Open Trade Sales over last mo.; increase over last yr.

Note 6 Accruais recorded for 12/31

Note 7 Member paid off receivable and the timing of the payment

Note 8 Decrease in Open Trade Purchases over last mo.; increase over last yr.

Note 9 Interim interest payments/adjustments and private market adjustments prior to period close

TCERA and TCERA Property, Inc. Combined Comparative Profit and Loss January 31, 2022

| | - | Jan 22 | Dec 21 | \$ Change | % Change | Jul '21 - Jan 22 | Jul '20 - Jan 21 | \$ Change | % Change |
|-----------|---------------------------------------|---------------|---------------|---------------|----------|------------------|------------------|---------------|----------|
| Ordina | ary Income/Expense | <i>يري</i> | | | | | | | |
| l. | ncome | | | | | | | | |
| | 4110 · Interest Income | 244,258.51 | 238,972.32 | 5,286.19 | 2.21% | 1,882,951.75 | 2,233,685.52 | -350,733.77 | -15.7% |
| (Note 1) | 4120 · Dividend Income | 156,906.78 | 344,312.43 | -187,405.65 | -54.43% | 1,966,598.11 | 2,284,299.30 | -317,701.19 | -13.91% |
| (Note 2) | 4130 · Real Assets Income | 1,653,787.99 | 0.00 | 1,653,787.99 | 100.0% | 3,500,679.33 | 2,896,573.65 | 604,105.68 | 20.86% |
| (Note 3) | 4140 · Other Investment Income | 37,047.20 | 3,274,849.50 | -3,237,802.30 | -98.87% | 4,131,506.07 | 7,194,361.51 | -3,062,855.44 | -42.57% |
| | 4200 · Lease Payments from TCERA | 15,640.00 | 15,640.00 | 0.00 | 0.0% | 109,480.00 | 109,480.00 | 0.00 | 0.0% |
| | 4310 · Commission Rebates | 0.00 | 241.06 | -241.06 | -100.0% | 458.88 | 1,931.88 | -1,473.00 | -76.25% |
| (Note 4) | 4410 · Securities Lending Income | 8,474.41 | 7,772.55 | 701.86 | 9.03% | 53,002.97 | 31,155.80 | 21,847.17 | 70.12% |
| (Note 5) | 4510 · Realized Gains/Losses | 7,941,681.19 | 8,942,718,27 | -1,001,037.08 | -11.19% | 83,358,784.71 | 50,066,463.40 | 33,292,321.31 | 66.5% |
| | 4530 · Gn/Ls Disposal of Fixed Asset | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| (Note 6) | 4610 · Employee Contributions | 990,952.93 | 2,992,457.82 | -2,001,504.89 | -66.89% | 14,111,447.78 | 12,942,330.81 | 1,169,116.97 | 9.03% |
| (Note 6) | 4620 · Employer Contributions | 115,409.46 | 463,707.77 | -348,298.31 | -75.11% | 40,927,851.10 | 37,288,643.61 | 3,639,207.49 | 9.76% |
| ٦ | otal Income | 11,164,158.47 | 16,280,671.72 | -5,116,513.25 | -31.43% | 150,042,760.70 | 115,048,925.48 | 34,993,835.22 | 30.42% |
| | | | | | | | | | |
| | Expense | | | | | | | | |
| | 5110 · Benefit Payments | 8,102,267.65 | 7,967,616.71 | 134,650.94 | 1.69% | 56,149,642.40 | 53,264,207.81 | 2,885,434.59 | 5.42% |
| (Note 7) | 5120 · Refunds | 752,829.32 | 1,908,072.62 | -1,155,243.30 | -60.55% | 4,466,642.53 | 1,726,796.89 | 2,739,845.64 | 158.67% |
| (Note 8) | 5130 · Death Retiree ROC | 18,129.92 | 0.00 | 18.129.92 | 100.0% | 31,638.15 | 76,646.62 | -45,008.47 | -58.72% |
| (| 5140 · SDA Payments | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| (Note 9) | 5210 · Investment Management Fees | 4,134.90 | 1,633,031.52 | -1,628,896.62 | -99.75% | 3,358,943.00 | 2,962,408.33 | 396,534.67 | 13.39% |
| (Note 9) | 5250 · Inv. Consultant/Custodial Fees | -80,134.51 | 202,558.32 | -282,692.83 | -139.56% | 210,743.89 | 198,793.84 | 11,950.05 | 6.01% |
| . , | 5270 · Securities Lending Expense | 704.24 | 716.87 | -12.63 | -1.76% | 4,740.67 | 5,009.20 | -268.53 | -5.36% |
| | 5275 · Real Assets Investment Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| (Note 9) | 5276 · Real Assets Mgr Fees | 0.00 | 609,574.12 | -609,574.12 | -100.0% | 1,098,584.31 | 842,733.31 | 255,851.00 | 30.36% |
| (Note 10) | 5280 · Other Investment Expense | 67,520.74 | 150,280.69 | -82,759.95 | -55.07% | 349,761.94 | 131,629.09 | 218,132.85 | 165.72% |
| (Note 9) | 5410 · Actuarial Study Fees | 8,500.00 | 50,838.75 | -42,338.75 | -83.28% | 112,527.89 | 113,866.50 | -1,338.61 | -1.18% |
| | 5450 · Compensated Benefit Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| * | 5500 · Administrative Expense | 156,614,65 | 233,465.79 | -76,851.14 | -32.92% | 1,290,959.22 | 1,199,047.54 | 91,911.68 | 7.67% |
| * | 5750 · TCERA Property Admin Expense | 9,745.41 | 3,147.14 | 6,598.27 | 209.66% | 34,015.45 | 24,350.12 | 9,665.33 | 39.69% |
| | 5910 · Depreciation of Fixed Assets | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| | Fotal Expense | 9,040,312.32 | 12,759,302.53 | -3,718,990.21 | -29.15% | 67,108,199.45 | 60,545,489.25 | 6,562,710.20 | 10.84% |
| Net O | rdinary Income | 2,123,846.15 | 3,521,369.19 | -1,397,523.04 | 39.69% | 82,934,561.25 | 54,503,436.23 | 28,431,125.02 | 52.16% |

TCERA and TCERA Property, Inc. Combined Comparative Profit and Loss January 31, 2022

| | Jan 22 | Dec 21 | \$ Change | % Change | Jul '21 - Jan 22 | Jul '20 - Jan 21 | \$ Change | % Change |
|--|----------------|---------------|----------------|----------|------------------|------------------|-----------------|----------|
| Other Income/Expense | | | | | | | | |
| Other Income | | | | | | | | |
| 4520 · Unrealized Gains/Losses | -41,134,538.78 | 24,394,641.02 | -65,529,179.80 | 268.62% | -61,071,610.72 | 146,199,762.54 | -207,271,373.26 | -141.77% |
| 4525 · Unrealized Gns/Ls Building/Land | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| 5000 · Other Income | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| | | | | | | | | ······ |
| Total Other Income | -41,134,538.78 | 24,394,641.02 | -65,529,179.80 | 268.62% | -61,071,610.72 | 146,199,762.54 | -207,271,373.26 | -141.77% |
| Other Expense | 0.00 | 0.00 | 0.00 | 0.0% | 0.00 | 0.00 | 0.00 | 0.0% |
| | | | | | | 1 | | |
| Net Other Income | -41,134,538.78 | 24,394,641.02 | -65,529,179.80 | 268.62% | -61,071,610.72 | 146,199,762.54 | -207,271,373.26 | -141.77% |
| Net Income | -39,010,692.63 | 27,916,010.21 | -66,926,702.84 | 239.74% | 21,862,950.53 | 200,703,198,77 | -178,840,248.24 | -89.11% |
| : | | | | | | | | |

Notes:

Note 1 Decrease in Dividends reported for Total Equity over last mo. and last year

Note 2 Real Estate Investment Income reported quarterly and one month in arrears

Note 3 Other Investment Income decrease over last mo. and last year

Note 4 Securities Lending Income increase over last year

Note 5 Realized Gains decrease over last mo.; increase over last year in Public Equity

Note 6 Accruals for EE & ER Contributions

Note 7 Semi-Annual Refund Accrual

Note 8 Increase in Retiree deaths with remaining contributions this mo; decrease over last year

Note 9 Accruals recorded for 12/31

Note 10 Decrease in Other Investment Expenses over last mo.:increase over last year in Priv. Credit

* See Budget report for detail of Administrative expenses (5500 and 5750)

TCERA and TCERA Property, Inc. BUDGET VS ACTUAL January 2022 Accounting Period 7

58.33% of the Current Fiscal Year Budget

| | | Jan 22 | Jul '20 - Jan 21 | Jul '21 - Jan 22 | Annual Budget | Remaining Budget | % Annual Budget |
|------------|--|-----------|------------------|------------------|---------------|------------------|-----------------|
| 5500 · Adı | ninistrative Expense | | | | | | |
| | 5505 · (6001) Allocated Salaries | 31,574.42 | 441,141.80 | 469,717.24 | 1,025,313.00 | 555,595.76 | 45.81% |
| | 5510 · (6002) Overtime | 0.00 | 0.00 | 0.00 | 1,000.00 | 1,000.00 | 0.00% |
| | 5515 · (6003) Other Pay | 2,027.54 | 5,239.31 | 9,546.02 | 28,672,00 | 19,125.98 | 33.29% |
| | 5520 · (6004) Benefits | 4,335.13 | 74,834.68 | 71,446.99 | 159,376.00 | 87,929.01 | 44.83% |
| (Note 1) | 5525 · (6005) Extra-Help | 920.40 | 0.00 | 8,381.99 | 500.00 | -7,881.99 | 1676.40% |
| | 5530 · (6006) Sick Leave Buy Back | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 | 0.00% |
| | 5535 · (6011) Retirement- Co. Port. | 4,015.09 | 56,313.66 | 60,439.78 | 136,235.00 | 75,795.22 | 44.36% |
| | 5540 · (6012) Social Security | 2,552.72 | 32,039,92 | 33,702,09 | 76,404.00 | 42,701.91 | 44,11% |
| | 5545 · (1024) POB Cost | 2,875.23 | 7,071.24 | 37,962.67 | 87,249.00 | 49,286.33 | 43.51% |
| (Note 2) | 5550 · (6008) Board Fees-Per Diem Pmts | 4,413.65 | 12,716.95 | 14,532.75 | 20,000.00 | 5,467,25 | 72.66% |
| | 5551 · Communications | | | | | | |
| | 5552 · (7005) Communications | 520.69 | 3,410.59 | 3,088.38 | 8,700.00 | 5,611.62 | 35,50% |
| | 5640 · (7005) Co. Telecommunication | 1,535.73 | 4,059.57 | 4,367.85 | 12,039.00 | 7,671.15 | 36.28% |
| | Total 5551 · Communications | 2,056.42 | 7,470.16 | 7,456.23 | 20,739.00 | 13,282.77 | 35.95% |
| | 5553 · Data Processing | | | | | | |
| | 5650 · (7044) ICT Qtrly - Data Process | 3,054.47 | 26,280.52 | 31,491.29 | 104,821.00 | 73,329.71 | 30.04% |
| | 5697 · (7044) Computer Exp Hdwr/Sftwr | 0.00 | 4,741.22 | 6,614.71 | 32,138.00 | 25,523.29 | 20.58% |
| | 5698 · (7044) WSI/Q2 Digital | 0.00 | 2,450.00 | 0.00 | 3,500.00 | 3,500.00 | 0.00% |
| | Total 5553 · Data Processing | 3,054.47 | 33,471.74 | 38,106.00 | 140,459.00 | 102,353.00 | 27.13% |
| | 5555 · (7009) Household Expense | 1,347.63 | 5,747.56 | 9,451.44 | 17,380.00 | 7,928.56 | 54.38% |
| (Note 3) | 5560 · (7010) Insurance | 0.00 | 54,402.00 | 58,624.00 | 59,400.00 | 776.00 | 98.69% |
| | 5565 · (7011) Unemployment Insurance | 0.00 | 0.00 | 14.67 | 6,000.00 | 5,985.33 | 0.00% |
| | 5570 · (7021) Maintenance-Equipment | 119.08 | 740.49 | 764.44 | 3,000.00 | 2,235.56 | 25.48% |
| (Note 4) | 5575 · (7027) Memberships | 1,500.00 | 6,010.00 | 6,050.00 | 7,100.00 | 1,050.00 | 85.21% |
| | 5580 · (7030) Due Diligence Expense | 0.00 | 0.00 | 2,521.92 | 15,000.00 | 12,478.08 | 16.81% |
| (Note 5) | 5585 · (7036) Office Expense | 9,741.05 | 9,126.11 | 32,663.34 | 50,080.00 | 17,416.66 | 65.22% |
| | 5586 · (7040) Courier | 236.64 | 556.77 | 708.78 | 1,460.00 | 751.22 | 48,55% |
| | 5590 · Prof & Specialized Exp | | | | | | |
| | 5591 · (7043) Prof & Special Gen Exp | 0.00 | 0.00 | 0.00 | 500.00 | 500.00 | 0.00% |
| (Note 6) | 5592 · (7043) Prof & Spec - Audit | 2,698.78 | 44,570.00 | 44,570.00 | 47,600.00 | 3,030.00 | 93.63% |
| | 5593 · (7043) Prof Sr -Outside Counsel | 16,315.30 | 11,726.70 | 25,371.10 | 75,000.00 | 49,628,90 | 33,83% |
| | Total 5590 · Prof & Specialized Exp | 19,014.08 | 56,296.70 | 69,941.10 | 123,100.00 | 53,158.90 | 56.82% |
| | 5594 · County Counsel Charges | | | | | | |
| | 5661 · (7046) Co Counsel - General Exp | 933.00 | 6,678.10 | 5,492.80 | 40,000.00 | 34,507-20 | 13.73% |
| | 5662 · (7046) Co Counsel - Disability | 758.60 | 33,992.00 | 20,814.80 | 93,000.00 | 72,185.20 | 22.38% |
| | Total 5594 · County Counsel Charges | 1,691.60 | 40,670.10 | 26,307.60 | 133,000.00 | 106,692.40 | 19.78% |
| | 5595 · (7049) Prof Exp-Disabilities | 8,291.20 | 34,565.00 | 33,869.28 | 164,000.00 | 130,130.72 | 20.65% |
| (Note 7) | 5600 · (7059) Publications | 9,090.91 | 12,567.85 | 13,466.27 | 22,000.00 | 8,533.73 | 61.21% |
| | 5675 · (7059) Co. Print Services | 4,215.80 | 2,444.91 | 4,757.50 | 14,500.00 | 9,742.50 | 32-81% |
| | Total 5600 · (7059) Publications | 13,306.71 | 15,012.76 | 18,223.77 | 36,500.00 | 18,276.23 | 49.93% |

TCERA and TCERA Property, Inc. BUDGET VS ACTUAL January 2022 Accounting Period 7

58.33% of the Current Fiscal Year Budget

| | | Jan 22 | Jul '20 - Jan 21 | Jul '21 - Jan 22 | Annual Budget | Remaining Budget | % Annual Budget |
|------------|--|------------|------------------|------------------|---------------|------------------|------------------|
| | | | Non- | Our Et Our EL | | Remaining Dudget | // Annuar Buuget |
| | 5605 · (7062) Rent & Lease -Building | 15,640.00 | 109,480.00 | 109,480.00 | 187,680.00 | 78,200.00 | 58,33% |
| | 5610 · (7066) Spec Dept Exp - RIS | 18,561.47 | 108,577.87 | 79,295.77 | 318,120.00 | 238,824.23 | 24.93% |
| | 5615 · (7073) Training | 415.25 | 1,970.00 | 2,245.25 | 15,500.00 | 13,254.75 | 14.49% |
| | 5620 · (7074) Transportation & Travel | 1,423.78 | 1,290.49 | 3,718.05 | 32,000.00 | 28,281.95 | 11.62% |
| (Note 8) | 5625 · (7081) Utilities | 1,455.02 | 10,701.10 | 12,481.67 | 20,400.00 | 7,918.33 | 61.18% |
| | 5627 · (7116) Postage - Co. Mail | 5,638.35 | 13,296.61 | 20,018.02 | 46,900.00 | 26,881.98 | 42.68% |
| (Note 3) | 5630 · (7128) Co. Workers Comp Insurance | 0.00 | 59,381.00 | 52,514.00 | 53,500.00 | 986.00 | 98.16% |
| | 5695 · Co. Admin. Services | | | | | | |
| | 5666 · (7719) HR/Risk Services | 0.00 | 0.00 | 0.00 | 11,860.00 | 11,860.00 | 0.00% |
| | 5699 · (7719) Auditors Services | 407.72 | 923.52 | 774.36 | 7,500.00 | 6,725.64 | 10.32% |
| | Total 5695 · Co. Admin. Services | 407.72 | 923.52 | 774.36 | 19,360.00 | 18,585.64 | 4.00% |
| | 5720 - (7421) Interest Expense | 0.00 | 0.00 | 0.00 | 1.00 | 1.00 | 0.00% |
| | 5450 - Compensated Benefit Expense | 0.00 | 0.00 | 0.00 | 26,353.00 | 26,353.00 | 0.00% |
| | 5911 · Depreciation - TCERA | 0.00 | 0.00 | 0.00 | 20,000.00 | 20,000.00 | 0.00% |
| | 5913 · Amortization - CPAS | 0.00 | 0.00 | 0.00 | 325,000.00 | 325,000.00 | 0.00% |
| Total TCE | RA Administrative Expense | 156,614.65 | 1,199,047.54 | 1,290,959.22 | 3,376,781.00 | 2,085,821.78 | 38.23% |
| 5750 · TCI | ERA Property Administrative Expense | | | | | | |
| | 5755 · Fees and Taxes | 0.00 | 25.00 | 25.00 | 200.00 | 175.00 | 12.50% |
| (Note 3) | 5760 · Insurance | 0.00 | 7,452.00 | 7,909.00 | 9,000.00 | 1,091.00 | 87.88% |
| | 5765 · Professional & Spec Services | 0.00 | 427.20 | 0.00 | 2,000.00 | 2,000.00 | 0.00% |
| | 5780 · Courtyards Property Assn Dues | 0.00 | 3,415.34 | 3,415.34 | 7,600.00 | 4,184.66 | 44.94% |
| | 5785 · Landscape Service | 450.00 | 2,422.25 | 2,655.00 | 7,800.00 | 5,145.00 | 34.04% |
| | 5790 · Security Monitoring | 806.00 | 1,368.16 | 2,236.03 | 7,750.00 | 5,513.97 | 28.85% |
| | 5795 · Maintenance & Improve - Bldg | 8,192.92 | 6,550.77 | 14,610.92 | 244,365.00 | 229,754.08 | 5.98% |
| | 5797 · Utilities | 296.49 | 2,689.40 | 3,164.16 | 7,000.00 | 3,835.84 | 45.20% |
| | 5912 · Depreciation - TCERA Property | 0.00 | 0.00 | 0.00 | 70,400.00 | 70,400.00 | 0.00% |
| Total TCE | RA Property, Inc. Administrative Expense | 9,745.41 | 24,350.12 | 34,015.45 | 356,115.00 | 322,099.55 | 9.55% |
| | RA & TCERA Property Administrative Expense | 166,360.06 | 1,223,397.66 | 1,324,974.67 | 3,732,896.00 | 2,407,921.33 | 35.49% |

Notes:

| Note 1 | Extra Help Office Assistant hired until full time replacement is found | Accrued Actuarial Liability | 1,957,985,000 |
|--------|--|-----------------------------|----------------|
| Note 2 | Quarterly Per Diem paid | June 30, 2021 | |
| Note 3 | Annual Insurance premiums paid | .21% of AAL | 4,111,768.50 |
| Note 4 | Annual memberships paid: SACRS, NCPERS, CALAPRS | | |
| Note 5 | Safety Election costs \$5,804 | 58.33% | 2,398,531.63 |
| Note 6 | Annual audit expense paid in the first half of fiscal year | Expenses to date | (1,324,974.67) |
| Note 7 | Qtrly newsletter & annual PAFR report | (Over)/Under | 1,073,556.96 |
| Note 8 | Increased electricity costs during summer months | | |



COUNTY OF TULARE BOARD OF RETIREMENT

Leanne Malison Retirement Administrator

136 N AKERS STREET VISALIA, CALIFORNIA 93291 TELEPHONE (559) 713-2900 FAX (559) 730-2631 www.tcera.org

TCERA, Board of Retirement Administrative Committee

Agenda Item # IV.2.b.

Agenda Date: March 23, 2022

Subject: TCERA's Disability Retirement Procedures Revision

Requests:

That the Administrative Committee:

- 1. Review the attached revised Disability Retirement procedures.
- 2. Forward the procedures to the Board of Retirement with a recommendation for approval.

Summary:

TCERA staff have prepared draft revisions to TCERA's Disability Retirement Procedures for the Committee's review. The recommended changes are designed to ensure that the information provided to disability applicants and their counsel, TCERA disability counsel, TCERA hearing officers, and TCERA trustees meets all legal requirements for disability application processing and is presented in a clear and complete format. Upon approval, the document will be included with each disability application packet and available to all interested parties. Please see the attached redline and final draft procedures for discussion regarding the proposed changes to the procedures.

Prepared by: Mary Warner

TULARE COUNTY EMPLOYEES' RETIREMENT ASSOCIATION (TCERA)

DISABILITY RETIREMENT PROCEDURES

Approved by the Board of Trustees on September 28, 2016 Amendments: May 10, 2017, May 13, 2020, April 13, 2022

1

1

TULARE COUNTY EMPLOYEES' RETIREMENT ASSOCIATION (TCERA) DISABILITY RETIREMENT PROCEDURES (Effective May 13, 2020)

PART I. GENERAL PROVISIONS

1. GENERAL PROVISIONS

1.1. Purpose

The purpose of these Procedures is to provide an equitable, fair and impartial method for acting upon applications for rights, benefits and privileges under the County Employees Retirement Law of 1937 (Gov. Code, § 31450 et seq.), to the end that applications may be expeditiously processed.

1.2. Definitions

In these Procedures, unless the context or subject matter otherwise requires:

- a. "Accommodations" means an agreement by the County or other participating employer to modify the regularly assigned duties and/or work environment of the member as necessary to allow the member to perform his/her regularly assigned duties, as modified, within his/her current job classification and within the limitations established by medical evidence.
- b. "Administrative record" means the record prepared by TCERA staff for proceedings in court, pursuant to Code of Civil Procedure section 1094.6.
- c. "Applicant" means: (1) a member of TCERA claiming benefits, rights, or privileges under the County Employees Retirement Law of 1937, (2) any person claiming such benefits, rights or privileges through a member, or (3) the County of Tulare or other participating employer in TCERA.
- d. "Association" or "TCERA" means the Tulare County Employees' Retirement Association.
- e. "Board" means the Board of Retirement.
- f. "Brief" means a written legal argument explaining why that side should win the case. References to factual matters contained in the record must include a citation to the volume and page of the exhibits where this information is shown.

- g, "County health officer" means the county health officer appointed by the Board of Supervisors pursuant to Health and Safety Code section 101000 et seq. (Pursuant to Gov. Code, § 24100, this term also includes any deputy.)
- h. "Day" means calendar day. The time to take an action specified under these Procedures shall be governed by general California law (see Code Civ. Proc., § 10 et seq.). For example, if the last day to file a document with the Association is a date when the Association office is closed prior to the end of regular business hours, or for the entire day (on a weekend or holiday or for some other reason), then the time period is extended to the next business day.
- i. "Disability" or "disabled" means permanent physical and/or psychological inability to substantially perform the regularly assigned duties of the job, with or without accommodations.
- j. "Effective date" of a disability retirement allowance means the date calculated pursuant to Government Code section 31724. As of the last date of amendment of these Procedures, this statute provides that the effective date of a disability retirement allowance can be:
 - 1) The date the disability retirement application is filed with the Board;
 - Not earlier than the day following the last day for which the member received regular compensation;
 - 3) After the expiration of any sick leave with compensation, unless the member agrees to retirement at an earlier date; or
 - 4) After the expiration of any leave of absence with compensation paid by the employer pursuant to Labor Code section 3201 et seq., unless the member agrees to retirement at an earlier date.

The statute also provides that when it has been demonstrated to the satisfaction of the Board that the filing of the member's application was delayed by administrative oversight or by inability to ascertain the permanency of the member's incapacity until after the date following the day for which the member last received regular compensation, such date will be deemed to be the date the application was filed.

- k. "Filed" means actually received by TCERA during regular business hours.
- *l.* "Formal hearing" means a hearing with live testimony by sworn witnesses, in addition to the hearing exhibits. Compare "hearing on the record."
- m. "Health care provider" means any person or entity which has provided medical or mental health care or treatment and has records which may be relevant to the disability retirement application, including but not limited to a physician, dentist,

psychologist, chiropractor, optometrist, physical therapist, acupuncturist, counselor, hospital, medical clinic, or laboratory.

- n. "Hearing exhibits" means the documents prepared by TCERA, plus those submitted by the parties, for a hearing.
- o. "Hearing officer" means an active member of the State Bar of California appointed by the Board pursuant to Government Code section 31533 to independently review the evidence and provide proposed findings of fact and recommendations to the Board, pursuant to the provisions of law and these Procedures.
- p. "Hearing on the record" means a procedure in which the hearing officer makes his/her recommendation based only on the hearing exhibits and any written legal arguments submitted by the parties (i.e., without live testimony). Compare "formal hearing."
- q. "Legal counsel" means an attorney who is an active member of the State Bar of California.
- r. "Member" means a member of TCERA.
- s. "Party" means any person disclosed by the records of the Association or by the application to have an interest in the subject matter of an application. Where appropriate within the context of the provision, this term may be limited to mean only the official participants in a hearing.
- t. "Pending" means the period from the time an application has been accepted as complete and ready for processing, through the final decision of the Board.
- u. "Permanent incapacity for performance of duty" means a permanent disability that precludes a member from substantially performing the regularly assigned duties of his/her job classification, even with accommodations.
- v. "Physician" means:
 - A person holding a valid and unrevoked physician's and surgeon's certificate or certificate to practice medicine and surgery, issued by the Medical Board of California or the Osteopathic Medical Board of California; or
 - A person with a doctoral degree in psychology, to whom a license has been issued pursuant to the provisions of California law, which license is in force and has not been suspended or revoked.
- w. "Retirement Administrator" means the individual employed as the Retirement Administrator by TCERA. The term includes a designee acting as Retirement

Administrator in the absence of the incumbent Administrator, as well as an Acting or Interim Retirement Administrator if the position of Retirement Administrator is vacant.

x. "Service" in certain contexts in these Procedures means that period of time for which deductions are made from a member's earnable compensation from the county or district. The "last day of service" is the last day of the last pay period for which retirement contributions are deducted. A member is considered to have discontinued service after this date.

For a member in a tier in which contributions end after thirty years of retirement service credit (Tiers 1-3), the "last day of service" may instead be the last day of the last pay period for which retirement contributions would have been deducted if the member did not have thirty years of retirement service credit. A member is considered to have discontinued service after this date.

- y. "Service" in other contexts in these Procedures means formally sending a document under these Procedures. "Proof of service" means a document certifying that a related document has been properly served on the required recipient(s) under these Procedures. Proper service means delivery of a document by one of the following methods:
 - 1) In person.
 - Depositing the document in first-class mail, postage pre-paid, addressed to the recipient at the address last known to the sender. Delivery is deemed complete when the document is mailed.
 - 3) Between TCERA and an employer which participates in the County interoffice mail system, by interoffice mail. Delivery is deemed complete when the document is picked up by the courier.
 - 4) With the written consent of the recipient, by email, facsimile, or other electronic means. Delivery is deemed complete when the document is sent, provided the sender does not receive notice that the document was not sent. (Note: TCERA has consented in these Procedures to receive electronic notice from the employer regarding counsel for the employer. See section 1.6.a.)
 - 5) Any document that is sent electronically after the close of business on that day shall be deemed to have been sent on the next business day.
- z. "Subrogation" means TCERA's right to recover retirement benefits paid to a member from a third party who caused the permanent incapacity of the member.

1.3. Burden of Proof

a. In an application for non-service connected disability retirement, the burden rests with the applicant to prove that he/she is permanently disabled from substantially performing the regularly assigned duties of his/her job.

Whenever an issue of drug or alcohol abuse is presented by the evidence, the burden set forth in this subdivision shall include the burden to prove that the particular incapacity was not the result of the alcohol or drug abuse.

- b. In an application for service-connected disability retirement, the burden rests with the applicant to prove that he/she is permanently disabled from substantially performing the regularly assigned duties of his/her job **and** that the employment contributed substantially to the disability.
- c. See Article 13 for the burden of proof rules where the employer is the applicant.

1.4. Communication with Individual Board Members

The Board acts as judge and jury regarding all disability retirement applications. As such, once an application for disability benefits has been filed, all communications between individual Board members and interested parties or their representatives regarding the application are forbidden until such time as the Board issues its written decision. Applicants are advised that any violation of this rule may result in the Board member being found ineligible to participate in any discussion regarding the application, or to influence the Board's decision, or to vote on the disability application.

- a. If a Board member receives a communication in violation of this provision, the Board member shall immediately take one of the following steps:
 - 1) If the communication is written, the Board member shall immediately provide to the Retirement Administrator a copy of the writing and any written response by the Board member to the communication.
 - 2) If the communication is oral, the Board member shall immediately send an email to the Retirement Administrator, stating the substance of the communication, any response made by the Board member, and the identity of each person from whom the Board member received the communication.
- b. The Retirement Administrator shall include the information provided by the Board member in the materials presented to the Board, and in the record. The Retirement Administrator shall notify the parties that the communication has been made part of the record.
- c. If a party requests an opportunity to address the Board regarding the communication within 10 days after the date of the notice from the Retirement Administrator, the Board may allow the party to do so.

1.5. Requirements for Board Member Participation

Any board member who has not reviewed the record that was provided to the entire Board for consideration shall not vote.

1.6. Right to Counsel

a. An applicant is entitled, at his/her expense, to be represented by legal counsel (as defined in section 1.2.q). An applicant may not be represented by a person who does not meet this definition.

Where the member is the applicant, the Association will recognize legal counsel as the representative of an applicant only upon receipt of a document signed by the applicant which lists the contact information for the attorney, including the attorney's California state bar number. Where the County is the applicant, the Association will recognize the appropriate team within the County Counsel's office as the representative of the applicant without documentation, or will recognize another legal counsel as the representative of an employer applicant upon receipt of any form of written notice from the employer (including an email) which lists the contact information for the attorney, including the attorney's California state bar number.

The Association reserves the right to contact the applicant directly in addition to or in lieu of contacting the applicant's legal counsel.

b. The Association is entitled to be represented by legal counsel, and shall be deemed to be represented by the appropriate team within the County Counsel's Office, unless otherwise designated by the Board.

1.7. Effect of Stipulations by Employer

The Board is not bound by any stipulation, agreement, understanding, act, or omission on behalf of the County of Tulare or other participating employer in TCERA by an officer, employee or agent of the County of Tulare or other participating employer in TCERA, in a workers' compensation or any other proceeding to which neither this Board nor the Association is a party, insofar as such stipulation, agreement, understanding, act, or omission may be asserted to relate to disability retirement or the disposition of a particular application for disability retirement.

1.8. Waiver of Procedures

a. The Board may waive or modify any provision of these Procedures, in whole or in part, to ensure the fairness of these Procedures for all parties, including the Association.

b. During an emergency which has been officially declared by an appropriate state official (as provided in Gov. Code, § 8680 et seq.), or by an appropriate County official (as provided in Tulare Co. Ord. Code, § 1-15-1000 et seq.), any party may request, and the Retirement Administrator may approve, an extension of any deadline in these Procedures until a reasonable period of time after the next scheduled meeting of the Board. If the Board is unable to take action at the next meeting as provided in section 1.8.a. for any reason (for example, losing a quorum or having a tie vote), or if the next meeting cannot be held until after the extension expires, the Retirement Administrator may continue to approve requests for extension until the Board is able to act. Both the requests and the approvals shall be in writing (email is acceptable).

1.9. Denial Due to Fraud or Willful Misrepresentation

If the applicant engages in any of the following actions, the application shall be denied with prejudice. Actions leading to denial on this ground include, but are not limited to:

- a. Willfully misrepresenting any fact or omitting relevant information in order to obtain disability retirement benefits.
- b. Presenting or causing to be presented any knowingly false or fraudulent written or oral statement in support of an application for disability retirement. For the purposes of this provision, the term "statement" includes records submitted to support the application.

1.10. Dismissal Due to Non-Compliance or Failure to Cooperate

In any matter in which the Association determines that the applicant has failed to comply with these Procedures, or to cooperate with the Association as reasonably required for the Association to process the application, the Association may dismiss the application without prejudice. The matter will be placed on a Board of Retirement meeting agenda for discussion and action regarding the dismissal. For purposes of this subsection, to "dismiss the application without prejudice" shall mean that the applicant may re-apply for a disability retirement whether or not the new application will be based upon the same facts and circumstances. If a new application is filed, the effective date of any disability retirement, if granted, will relate to the date of the new application as required by statue, and not to the date of the application which was dismissed.

1.101. Withdrawal

- a. An applicant may withdraw an application at any time prior to the initial decision of the Board.
- b. An applicant may withdraw a request for hearing at any time between the initial decision of the Board and the final decision of the Board.
 - 1) The withdrawal must be in writing and must be filed with the Association.

Formatted: Indent: Left: 0", First line: 0'

2) The withdrawal is irrevocable once filed, and makes the initial decision of the Board the final decision. The decision cannot be appealed to court because without a hearing the applicant has not exhausted administrative remedies.

1.1+2. No Additional Time for Mailed Documents

The extension of time provided by Code of Civil Procedure section 1013 when a document is served by mail does **not** apply to any deadlines set by these Procedures.

PART II. APPLICATION THROUGH INITIAL BOARD DECISION

2. APPLICATION FILED BY MEMBER¹

2.1. Furnishing Copy of Procedures

The Association shall include a copy of these Procedures in the packet provided to anyone wishing to file an application. The Association shall also provide a copy to any member of the public on request, or, if these Procedures are posted on the TCERA website, may direct the requesting person to the website.

2.2. Filing Application

The Association shall furnish application forms for disability retirement benefits upon request to any person authorized to file an application. The applicant shall fully complete the application forms in accordance with these Procedures and the instructions stated on the application forms, and shall sign and verify the completed forms under penalty of perjury.

The completed application shall be filed with the Association. Any application or document filed in support thereof that is incomplete or illegible shall not be accepted for filing, but instead shall be returned to the applicant.

2.3. Documents to Be Included in Application

The following documents and reports shall be completed or furnished by the applicant in connection with the initial application. All documents must be legible and capable of being photocopied legibly.

- a. Completed Application for Disability Retirement.
- Responses to Disability Retirement Questionnaire, signed under penalty of perjury. (See section 1.9 regarding misrepresentation or omission of relevant information.)

¹ See Article 13 for provisions regarding employer-filed applications.

The applicant shall list **all** health care providers seen by the applicant beginning with the date which is five (5) years prior to the date of the injury, diagnosis, or other disabling condition, and continuing through the date of application. The applicant shall also list all providers seen earlier than this five (5)-year date regarding the same or a similar injury, diagnosis, or condition, and shall include the approximate dates seen.

The applicant shall sign and attach to his/her responses the signature page which the Association provides with the Questionnaire which includes the declaration under penalty of perjury. If the responses are not returned with the appropriate signature page they shall be deemed not to have been filed and shall be returned to the applicant with additional instructions.

- c. Completed release forms to enable the Association to obtain the Applicant's medical and personnel records.
- d. Completed Treating Physician Statement(s).

The Treating Physician Statement (TPS) must state that the member is permanently incapacitated for the performance of the duties of his/her position, and, if the application is for service-connected disability retirement, that the member's employment contributed substantially to such incapacity. The TPS must be signed by a_physician (as defined in these Procedures) with expertise in the type of disability claimed by the member.

If the TPS does not state that the member's employment contributed substantially to the incapacity, the application will be treated as one for non-service-connected disability retirement, even if the applicant has stated in the application that he/she believes the disability is directly related to employment with an employer participating in TCERA.

e. Additional and supplemental Releases and Questionnaires may be required in the course of review of the application.

2.4. Basis for Application

A disability retirement application shall be based upon a permanent medical condition which constitutes a physical or psychological incapacity, or a combination of both. The documents on file must clearly identify the exact nature of the incapacity that forms the basis for the disability application. A disability retirement will not be granted except for those incapacities stated in the application on file and proven by the applicant to the satisfaction of the Board.

2.5. Time Limit for Filing

- a. An application for disability retirement shall not be deemed effective or filed within the meaning of Government Code sections 31721 through 31724, inclusive, until the date on which it is received by the Association at the office indicated in Section 2.6. An amended application shall be deemed filed, within the meaning of these statutes, on the date it is received by the Association. It is the applicant's responsibility to obtain such documents and the Association shall not process any disability retirement application until all the information required in Section 2.3 has been furnished, at which time the application shall be deemed complete and ready for processing.
- b. The burden rests with the applicant to establish timeliness of filing. As required by Government Code section 31722, the application must be filed:
 - 1) While a member is in service;
 - 2) Within four months after the member's discontinuance of service;
 - Within four months after the expiration of any period during which a presumption is extended beyond the member's discontinuance of service; or
 - 4) While, from the date of discontinuance of service to the time of the application, the member is continuously physically or mentally incapacitated to perform his/her duties.

Applicants should note that if the application is filed under this ground, the applicant has an additional burden regarding timeliness (proving continuous disability) which does not apply to the first three grounds.

- c. See definition of "service" in section 1.2.x.
- d. The applicant may show that the member is entitled to an effective date of disability retirement earlier than the date of the application, pursuant to Government Code section 31724. (See definition of "effective date" in section 1.2.j.)

2.6. Where to File

Application forms and related materials such as these Procedures may be obtained from the Association, 136 North Akers Street, Visalia, California 93291. All forms and documents shall be filed in hard copy at this location unless otherwise specified. All telephone inquiries should be made to (559) 713-2900.

2.7. Amendment of Application

- a. An applicant may amend his/her application at any time before the Board's initial decision. The new information shall be added to the staff review and recommendation to the Board.
- b. After the Board's initial decision, an application may not be amended. The applicant must instead file a new application, which will be held without investigation until the pending application has been finally decided, and may then file a new application. The effective date of benefits pursuant to the new application shall be determined as if no prior application had been filed.

3. STAFF REVIEW AND RECOMMENDATION

3.1. Review of Application

Once all required information has been submitted and the application is deemed complete and ready for processing, the Association shall process the application as set forth below.

3.2. Records in Support of Claim

- a. The applicant shall make arrangements for all medical records from the health care providers listed in the responses to the questionnaire to be sent directly to the Association by the physician, medical clinic, hospital, or other health care provider. Records submitted directly by the applicant will be rejected, unless the health care provider seals the records in a manner acceptable to the Association.
- b. The applicant may submit a written request to the Association to be excused from providing particular records, for good cause. The request shall be signed under penalty of perjury. The Association in its sole discretion may grant or deny the request, and the response shall also be in writing. The Association may provide a form for the request.
- c. The Association at any time may demand that the applicant arrange for additional records which the Association believes may be relevant to the application to be provided to the Association. Disputes regarding such records may be referred to the Retirement Administrator. The Association may also require the Applicant to provide additional and supplemental releases and responses to questions in the course of review of the application.

3.3. Medical Examinations

a. A member may be required to submit to medical examinations by a physician or other expert identified by the Association. The Association shall pay for the cost of any such examination. Refusal of any applicant to submit to such medical examination shall be grounds for considering that the results of such examination would be unfavorable to the applicant, and such refusal shall result in denying the application with prejudice. The cost for an Independent Medical Examination or review estimated to exceed the amount set from time to time by the Board shall be approved in advance by the Board.

- b. The Association will reimburse vehicle mileage to and from a medical examination outside Tulare County, as well as reasonable lodging and meals at a rate not greater than the County per diem amount, and as authorized by the County reimbursement policy, **only** when the medical examination has been at the request of the Association. Mileage reimbursement will be made at the County mileage rate after the member has attended the appointment, for the number of miles from the member's residence directly to the examination location, as determined by Association staff from a mapping service on the Internet, rounded to the nearest whole mile. Reimbursement for meal costs may also include those for a companion where one is needed to transport the applicant to the site of the medical examination, but only when approved in advance by the Association.
- c. If the applicant fails without sufficient excuse to attend an examination scheduled at the request of the Association, the applicant shall reimburse the Association for any missed appointment fee charged by the examining physician. The determination of what constitutes a sufficient excuse shall be within the sole discretion of the Board. Failure to pay the reimbursement in full, or to make acceptable arrangements to pay it in installments, shall constitute cause to dismiss the application, with or without prejudice as determined by the Board.
- d. All costs incurred for any and all medical examinations, medical reports and other charges which are undertaken or prepared at **the request of the member** shall be borne solely by the member.

3.4. Other Investigation

The Association may hire third parties, including but not limited to a private investigator, to investigate the issues and/or claims made by a disability retirement applicant that relate to his/her entitlement to receive disability benefits.

3.5. Information from Employer

At any time during the review process, the Association may request that the applicant's employer provide information regarding the member's job duties, illness or injury, accommodations, or other pertinent data. Additional reasonable requests may be made at any time as deemed necessary by the Association. Requests shall include a release by the applicant or other appropriate documentation. A participating employer shall provide the underlying data, or a response if requested by the Association, in writing, no later than 30 days after the date of the request. The Retirement Administrator may grant an extension of time of up to 10 additional days to respond upon a showing of good cause by the employer.

3.6. Submittal to Board

After obtaining all information deemed to be relevant and necessary, staff shall submit the matter to the Board.

4. INITIAL BOARD DECISION

4.1. Advice from County Health Officer

As provided in Government Code section 31530, the Board may seek advice from the county health officer at any time, and, if requested by the Board, the county health officer shall attend the Board's meetings.

4.2. Board's Options

Upon review of an application for disability retirement, the Board may do any of the following:

- a. Grant the benefit applied for.
- b. If the applicant has applied for service-connected disability retirement, and is eligible for non-service-connected disability retirement, grant a non-service-connected disability retirement, and:
 - Deny the application for service-connected disability retirement, and direct staff to inform the applicant of his/her right to request a hearing on the issue of service connection, or
 - 2) Refer the issue of service connection to a hearing for a determination.
- c. Refer the entire matter to a hearing for a determination on any or all issues presented.
- d. Deny the application in its entirety and direct staff to inform the applicant of his/her right to request a hearing.
- e. Refer the matter back to staff with instructions.
- f. Dismiss the application without prejudice.
- g. Dismiss the application with prejudice.

4.3. Board Decision

Each decision shall be made by and shall be the decision of the Board. Each such decision shall be based upon all of the information presented to the Board.

In the event the Board is unable to act, in whole or in part, due to lack of a majority vote in accordance with the Board's bylaws, the minutes shall indicate that pursuant to these Procedures, the inability to take action is deemed an action to bring the matter back at the next regular meeting. If the Board is still unable to act at the second meeting, the minutes shall indicate that the inability to take action is deemed an action to deny the application, in whole or in part as applicable, pursuant to these Procedures.

PART III.

HEARING PROCESS AND FINAL BOARD DECISION

5. PROCEDURES APPLICABLE TO ALL HEARINGS

5.1. Referral by Board to Hearing for Determination

- a. If the Board's decision is to refer the matter to a hearing for a recommendation, the Board shall assign the matter to a hearing officer. The applicant and the applicant's employer shall be notified within ten (10) days of the Board's action. The Association's counsel shall be concurrently notified by copy of said notice.
- b. The notice shall advise that the matter will be determined by a hearing on the record, unless a formal hearing is requested by any party.
- c. A written request for a formal hearing may be filed by any party within ten (10) days after the date of the notice. (See section 5.2.c. regarding requirements for requesting a formal hearing; see also section 1.11 regarding no extension of time if a document is mailed.)

5.2. Request for Hearing after Denial of Application by Board

- a. If the Board's decision is to deny the application in whole or in part, the member, and the member's employer, shall be notified within ten (10) days of the Board's decision. The applicant and the Association shall have the right to request a hearing. The party requesting a hearing may request either a hearing on the record, as provided in Article 6, or a formal hearing, as provided in Article 7. A formal hearing will be scheduled if requested by any party.
- b. A request for hearing **must** be filed with TCERA not later than ten (10) days after the date of the notice of denial. (See section 1.11 regarding no extension of time if a document is mailed.)
- c. If the request is for a formal hearing, the party must indicate on the included calendar any dates which are unacceptable for a hearing. The party must also indicate the expected number of hearing days required to hear the matter. When a

request for a formal hearing has been received, the Association shall immediately forward a copy to the non-requesting party, who shall within ten (10) days after the date of the notice indicate on the included hearing calendar any unacceptable dates. The parties are responsible for clearing calendars for any medical or other witnesses prior to the time that the responsive calendar is submitted. See section 7.3.c. (disallowed grounds for postponement or continuance).

If the applicant desires legal counsel, it is the applicant's responsibility to obtain counsel sufficiently in advance of the hearing date to allow counsel adequate time to prepare for the hearing. See section 7.3.c (disallowed grounds for postponement or continuance).

A failure of any party to submit a calendar indicating the dates not acceptable for hearing shall be deemed consent to set the hearing on any weekday indicated to be available on the calendars that were sent to both parties.

5.3. Changing Type of Hearing after Initial Selection

- a. A party who has requested a hearing on the record may not request that the matter be changed to a formal hearing.
- b. A party who has requested a formal hearing may request changing to a hearing on the record. Provided the other party has not requested a formal hearing, and/or stipulates to changing to a hearing on the record, the Retirement Administrator may grant the request.

The request must be filed with the Association not later than 30 days before the date the formal hearing is initially scheduled to begin.

If the Retirement Administrator grants the request, the triggering event for purposes of the 30-day period to file briefs as provided in section 6.2 shall be either the date of service of the approval, or the date set forth in section 6.2, whichever is later.

5.4. Assignment of Hearing Officer

- a. Once a hearing has been requested or ordered, the Association shall assign the next available hearing officer in the rotation to the hearing. The Association shall notify the parties promptly when a hearing officer has been assigned.
- b. Each party shall be entitled to request reassignment of the hearing to another hearing officer without cause. Each party shall be entitled to make only one (1) such request.
 - 1) The request must be in writing and must be filed with the Association no later than 10 days after service of the notice of assignment of hearing officer.

- 2) A party which did not request reassignment the first time may file its own request for reassignment as described above after the notice of assignment of the second hearing officer is served.
- c. At any time, a party may request reassignment to another hearing officer for good cause.
 - 1) The reassignment request shall be supported by a declaration under penalty of perjury, and shall be made in the form provided in Appendix A.
 - 2) Upon the filing of a request for reassignment in accordance with the provisions of this section 5.4.c., without any further act or proof, the Association shall reassign the case to the next available hearing officer in the rotation.
 - 3) Upon reassignment, a new notice of assignment of hearing officer shall be served on the parties. If there are no available hearing officers left in the rotation, the Association shall take such steps as are appropriate under the circumstances, which may include contracting with the Office of Administrative Hearings to provide an administrative law judge.

5.5. Authority of Hearing Officer

The hearing officer may require the parties to hold teleconferences as needed.

The hearing officer may also make any procedural rulings needed, and, upon the stipulation of all parties, the hearing officer may waive or modify any provision of the rules in Article 6 or 7 for that type of hearing, in whole or in part, except for the time limits in section 7.3.

The hearing officer shall not engage in ex parte communications with a party. Discussions with one party about scheduling the hearing do not constitute ex parte communications.

5.6. Initial Exhibit Packet

For all hearings, the Association shall prepare the initial exhibit packet as follows.

a. Assembly of Packet

Within 45 days after the notice of initial assignment of the hearing officer, the Association shall prepare and serve a consecutively numbered, tabbed exhibit packet consisting of all records provided by the applicant or the applicant's representative and/or the employer or employer's representative, and all records obtained by the Association in the application process. The initial exhibit packet shall consist only of those records in the Association's possession as of the date of the initial Board decision.

b. Service of Packet

The Association shall provide electronic copies of the initial exhibit packet to all parties, with a proof of service indicating the date of distribution. The Association shall also provide hard copies for the hearing officer and TCERA's counsel at no charge, and shall provide a hard copy for the applicant on request upon payment of the fee for preparation of a hard copy.

5.7. Supplemental Exhibit Packet

The Association shall provide a supplemental exhibit packet (if any), or a notice of no supplemental exhibit packet, as follows:

- a. Within 45 days after the date of service of the initial exhibit packet, a party shall do one of the following:
 - 1) Advise the Association in writing that the party will not submit any additional records for a supplemental exhibit packet.
 - 2) Advise the Association in writing that the party may wish to submit additional records for a supplemental exhibit packet.

This notice may include a written request to the Association to issue subpoenas for records to be provided directly to the Association, in circumstances where the party is legally barred from obtaining the records, or where the party shows the Association that reasonable efforts to obtain the records have been unsuccessful.

Neither notice needs to be served on the other party or parties.

- b. If a party wishes to submit additional records for a supplemental exhibit packet, that party must arrange for the additional records, whether subpoenaed or not, to be provided directly to the Association not later than 135 days after the date of service of the initial exhibit packet as described in section 5.6.b. (i.e., not later than 90 days after the deadline in section 5.7.a.).
- c. If all parties advise pursuant to section 5.7.a.1. that no additional records will be submitted, then, within 15 days after the date of receipt of the last notice, or the date in section 5.7.a., whichever is earlier, the Association shall notify the parties, with a proof of service, that no supplemental exhibit packet will be provided.
- d. If one or more parties advise the Association that additional records may be submitted, then within 15 days after the deadline in section 5.7.b., the Association shall prepare a consecutively numbered, tabbed supplemental exhibit packet consisting of all records provided by the deadline to the Association, and shall provide it to all parties and the hearing officer, with a proof of service indicating the date of distribution, in the same manner as in section 5.6.b.

5.8. Objections to Records in Packets

All records in the initial exhibit packet and the supplemental exhibit packet, if any, shall be deemed to be properly authenticated, and shall not be treated as hearsay unless the objecting party, not later than thirty (30) days after delivery of the initial or supplemental exhibit packet, files with the hearing officer, with a proof of service on the other party or parties, a request to exclude certain records, with a description detailing the basis for the objection with sufficient specificity. Nothing in this rule precludes a party from arguing about the substance of, or challenging the truth of, any material contained in the exhibit packet or packets as prepared by the Association. The rules of privilege shall be effective to the extent that they are otherwise required by statute to be recognized at the hearing.

5.9. Submittal of Additional Records

- a. No additional records may be submitted to the Association after the date of service of the supplemental exhibit packet, or notice of no supplemental exhibit packet, as described in section 5.7.
- b. The hearing officer may allow additional medical reports or other documents to be submitted directly to the hearing officer after the date in section 5.7. The requesting party must show that such records are directly relevant to an issue in dispute, and that the evidence is not cumulative. The hearing officer must permit submission in any of the following circumstances:
 - 1) Where records that were *timely* subpoenaed were not received by the Association in time to be included in the supplemental exhibit packet.
 - 2) Where records *which predate the date in section 5.7.a.* are discovered which the party neither knew about nor should have known about, in the exercise of reasonable diligence, and where the party proposing to submit such records provides notice to the other party or parties within 24 hours after learning of the existence of such records, and provides copies to the other party or all other parties within 24 hours after obtaining the additional records.
 - 3) With the consent of the Retirement Administrator.
- c. If the hearing officer permits submission under section 5.9.b., the party shall provide copies at no cost to the hearing officer and all other parties. The hearing officer may determine timelines for submission of documents not covered by section 5.9.b.2., and/or may set or reset other deadlines in Article 6 or 7 as needed, except those in section 7.3.
- d. In a hearing on the record, the hearing officer must reject any documents which do not meet the criteria in section 5.9.b.

e. In a formal hearing, the hearing officer may permit submission of documents which do not meet the criteria in section 5.9.b. See section 7.12 (Submittal of Additional Records at Hearing).

5.10. Additional Facts Arising Before Hearing

If facts which could have affected the Board's initial decision, but were not known at the time of that decision, come to light before the hearing commences, any party, or staff, may bring these facts to the attention of the Retirement Administrator. The Retirement Administrator may determine that these facts are material enough that they should be presented to the Retirement Board.

If that determination is made, the Retirement Administrator shall cancel any hearing process and shall return the matter to the Board for further review pursuant to Article 4 (Initial Board Decision). If the Board again denies the application, in whole or in part, the applicant may again request a hearing.

If a new request for hearing is submitted, the matter will be heard by the same hearing officer as assigned earlier, unless he/she is not available, in which case all procedures in section 5.4 (Assignment of Hearing Officer) will be followed.

6. HEARING ON THE RECORD

6.1. Exhibit Packet

The rules for the exhibit packet are set forth in Article 5. As noted in section 5.9.d., the hearing officer must reject any supplemental documents which do not meet the criteria set forth in section 5.9.b.

6.2. Briefs

Not later than thirty (30) days after the date of service of the supplemental exhibit packet, or notice of no supplemental exhibit packet, as described in section 5.7, the parties may file simultaneous written briefs (as defined in section 1.2.f.) with the Association, with a proof of service on the hearing officer and the other party or parties, which identifies issues, evidence, and relevant law, and presents arguments. (For help in writing a brief, parties may search the Internet for "how to write a legal argument in a court case.") If a brief claims that something is in the record, the brief must identify the volume and page of the record where the item appears.

One original of each brief must be filed in hard copy with the Association. The parties may stipulate to serve solely electronic copies on each other, and the hearing officer may consent to be served solely with an electronic copy.

The hearing officer may, for good cause, grant a request to extend time for the parties to file their simultaneous briefs. The request must be made no later than seven (7) days before

the original deadline, and the new deadline shall be no later than 15 days after the original deadline. The hearing officer shall notify the Association and the parties in writing whether the request for an extension has been granted or denied. If the request is granted, the deadline for the hearing officer's proposed findings of fact and recommendations pursuant to section 6.3 below shall be extended accordingly.

6.3. Hearing Officer's Recommended Decision

The hearing officer shall review the record and any briefs, and shall submit his/her proposed Findings of Fact and Recommendations to the Board of Retirement, with a proof of service on the parties, no later than 30 days after the date the briefs were filed or due, as described in section 6.2.

The hearing officer shall include in his/her recommended decision a notice to the parties that they have ten (10) days from the date of service of the recommended decision to submit written objections, with a proof of service on the other parties, to the Board.

6.4. Objections to Hearing Officer's Recommended Decision

Pursuant to Government Code section 31534, the parties shall have ten (10) days from the date of service of the hearing officer's proposed Findings of Fact and Recommendations to submit written objections, with a proof of service on the other parties, to the Board. (The objections are **not** served on the hearing officer.) These objections shall be incorporated in the record to be considered by the Board when it considers the proposed decision.

7. FORMAL HEARING

7.1. Initial Scheduling of Hearing

The hearing must initially be scheduled to commence within one hundred eighty (180) days after the date of service of the initial exhibit packet, as described in section 5.6.

The Association or its counsel shall initiate the scheduling process. The parties are expected to cooperate in setting the hearing date or dates. The hearing officer shall settle disputes regarding the initial scheduling of hearing dates.

The hearing may be postponed as provided in section 7.3.

Hearings that are not concluded on the date set may be continued to any future date agreed to by all parties, provided the new date meets the requirements of section 7.3.

7.2. Notice of Hearing

Once the initial hearing date has been established, the Association shall promptly issue a formal notice of hearing to each party describing the time and place of the hearing. Such notice shall be provided not later than forty-five (45) days prior to the initial hearing date.

7.3. Postponements and Continuances

a. Definitions

- 1) "Postponement" means delaying the beginning of a hearing prior to the submission of any formal evidence or testimony.
- "Continuance" means delaying resumption of a hearing after the hearing has begun, i.e., after some formal evidence and/or testimony has been received from one or more of the parties.

b. Limitations

- 1) The hearing officer may grant one request by each party to postpone the hearing for good cause, and may also grant one request by each party to continue the hearing for good cause. Any request for postponement or continuance shall be made not later than fifteen (15) days in advance of the next scheduled date of the hearing, except when the basis for the postponement or continuance is not known fifteen (15) days in advance. The hearing officer shall promptly grant or deny any request.
- 2) Only the Retirement Administrator may grant additional requests by a party for postponement or continuance for good cause. The Retirement Administrator must also approve treating a postponement or continuance as not charged to any party, such as when the hearing officer requests a postponement or continuance on his/her own behalf.
- 3) Any postponement or continuance must be to a date certain which is within 45 days after the then currently scheduled date, or the most recent date on which the hearing was actually held, as applicable.
- 4) If the request is granted, the hearing officer shall indicate in writing whether the postponement or continuance is charged to a party, and, if yes, which party or parties have used their requests and which still have a request available.

c. Good cause

Good cause shall be within the discretion of the hearing officer or Retirement Administrator, as applicable. However, the following shall **not** constitute good cause for postponement or continuance of a hearing:

 The desire to await the outcome of an application for workers' compensation or to exhaust benefits under Labor Code section 4850 and/or long term disability pay.

- 2) The need to permit a condition to stabilize in order to determine whether the disability is permanent.
- 3) Failure to obtain legal counsel, if so desired, sufficiently in advance of the hearing date to allow counsel adequate time to prepare for the hearing.
- 4) Conflicting schedule of a witness, where that conflict was known at the time the hearing calendar was submitted.

d. Hearing officer's authority

The hearing officer shall settle disputes regarding the scheduling of postponed or continued hearing dates.

7.4. No Discovery

No discovery requests to other parties are allowed, including depositions, requests for production of documents, requests for admission, and all other discovery permitted in civil cases, except as expressly permitted in these Procedures.

7.5. Submission of Witness Lists

Not later than thirty (30) days before the scheduled hearing date, all parties or their counsel shall submit to the hearing officer, with a proof of service on the other party or parties, a list of the names of any physicians or other witnesses who will or may be called to testify.

If the hearing is postponed or continued, the hearing officer may require a supplemental list to be submitted with such deadlines and other requirements as the hearing officer may set.

7.6. Independent Medical Examination

Not later than thirty (30) days before the scheduled hearing date, a party may request, or the parties may stipulate to request, that the Retirement Administrator authorize an independent medical examination. The request must be in writing and must show good cause to grant the request. The Retirement Administrator may grant the request if good cause is shown.

7.7. Subpoenas for Attendance of Witnesses at Hearing

It shall be the responsibility of each party to obtain those subpoenas for witnesses to appear at the hearing that he/she deems necessary for the presentation of his/her respective portion of the case. The party requesting the subpoenas must submit to the Association a written request with the name of each person to be subpoenaed. The parties are strongly advised to request subpoenas as soon as the party knows the witness will be called. The Association shall provide the completed subpoena forms within **five (5) working days** after such request. Subpoenas shall be signed by the Chair of the Board, the Retirement Administrator, or the hearing officer.

Neither the Board nor the Association is responsible for the service of subpoenas, the cost of service, or the payment of witness fees. All subpoenas are to be issued in accordance with Government Code section 31535. The parties and their counsel are responsible for complying with all legal requirements regarding the request for and service of subpoenas and regarding any other laws which may impact such subpoenas.

7.8. Affidavits

- a. At any time twenty (20) or more days prior to a hearing or a continued hearing, any party may mail or deliver to the opposing party a copy of any affidavit that he/she proposes to introduce, as set forth below. Unless the opposing party, within ten (10) days after such mailing or delivery, mails or delivers to the proponent a request to cross-examine an affiant, his/her right to cross-examine such affiant is waived and the affidavit, if introduced in evidence, shall be given the same effect as if the affiant had testified orally. If an opportunity to cross-examine an affiant is not afforded after request therefor is made as herein provided, the affidavit shall be excluded.
- b. The notice referred to in subdivision a. shall be substantially in the following form:

The accompanying affidavit of (here *insert name and contact information of affiant*) will be introduced as evidence at the hearing in (here *insert title of proceeding*). (Here *insert name of affiant*) will not be called to testify orally and you will not be entitled to question him/her unless you notify (here *insert name of proponent or his/her attorney*) at (here *insert address*) that you wish to cross-examine him/her. To be effective your request must be mailed or delivered to (here *insert name of proponent or his/her attorney*) on or before (here *insert a date 10 days after the date of mailing or delivering the affidavit to the opposing party*).

7.9. Pre-Hearing Briefs

A party may, but is not required to, submit a pre-hearing brief. (This is most often done when a case presents a technical, often legal, issue.) A party wishing to submit a prehearing brief shall notify the hearing officer and other parties no later than 14 days prior to the scheduled date of the hearing of the party's intent to submit a pre-hearing brief. One original of a pre-hearing brief must be filed in hard copy with the Association. Copies must be served on the hearing officer and the other party or parties in such a manner that they receive it no later than two (2) days prior to the hearing, unless otherwise ordered by the hearing officer. The other party or parties and the hearing officer may consent to be served solely with an electronic copy.

7.10. Stenographic Reporter

The Association shall arrange for a certified shorthand reporter to record the hearing at the expense of the Association.

7.11. Rights of Parties

Oral evidence shall be taken only on oath or affirmation. Each party shall have these rights: to call and examine witnesses; to request that additional records be accepted; to crossexamine opposing witnesses on any matter relevant to the issues even though that matter was not covered in the direct examination; to impeach any witness regardless of which party first called him/her to testify; and to rebut the evidence against him/her. If the applicant does not testify, he/she may be called and examined as if under crossexamination. At the request of any party, witnesses may be excluded from the hearing, except while testifying.

7.12. Submittal of Additional Records at Hearing

- a. The hearing officer may permit additional records to be submitted at the hearing. The requesting party must show why the records should be permitted despite not meeting the criteria in section 5.9.b.
- b. Should the hearing officer permit submission of records, he/she shall condition the submission upon such terms as he/she deems fair, including without limitation: permitting the non-submitting party to subsequently file a declaration of other witnesses regarding the newly submitted evidence; permitting the non-submitting party to call the witness who authored the evidence at the expense of the submitting party; permitting the non-submitting party to depose the witness at the expense of the submitting the hearing to permit time to properly respond to the new records, which, if granted, shall not be counted as the non-submitting party's one allowable request to the hearing officer. Additionally, the hearing officer may consider the records as offered but afford them only weight as uncorroborated hearsay.

Prior to ruling on the admissibility of any belatedly presented records, in order for the submitting party to determine if he/she wishes to pursue the issue, the hearing officer shall advise the submitting party of the possible conditions to admissibility which might be imposed. If the submitting party does pursue submission and the evidence is later admitted, it shall be conclusively presumed that the submitting party consents to the conditions identified by the hearing officer regarding the admission of the evidence.

c. A party wishing to submit additional documents shall bring enough copies to the hearing to assure that the hearing officer and all parties receive a copy.

d. The hearing officer shall ensure that any documents submitted are marked as additional exhibits, whether admitted into evidence or not, and shall forward the hearing officer's copies to the Association after the hearing for inclusion in the record as required by Code of Civil Procedure sec. 1094.6 ("Such record shall include ... all rejected exhibits in the possession of the local agency or its commission, board, officer, or agent").

7.13. Rules of Evidence

- a. Formal rules of evidence do not apply, but evidence may be admitted only if it is the kind of evidence upon which reasonable persons are accustomed to rely in the conduct of serious affairs. Hearsay evidence alone shall be insufficient to support a decision. As provided in section 5.8, records included in the exhibit packet(s) prepared by the Association are not hearsay unless a timely objection is made.
- b. Any objections to the admissibility or relevance of evidence other than evidence admitted pursuant to section 5.9.b. shall be determined by the hearing officer. Notwithstanding this rule, any party may object at the hearing that such documents are, in whole or in part, irrelevant.
- c. The hearing officer shall retain the right to exclude evidence which is not relevant or which is cumulative, or otherwise found not probative.
- d. Any party may request that the hearing officer take official notice of those matters provided for in Evidence Code sections 451 through 453.
- e. Other than rebuttal witnesses, neither party may call as a witness any physician or other witness whose name has not been disclosed to the other party pursuant to the provisions of section 7.5 (Submission of Witness Lists), except as provided in section 7.12.b. (possible conditions for submission of additional records at hearing).

7.14. No Additional Records after Conclusion of Hearing

No additional records may be submitted after the conclusion of the hearing. The hearing concludes when all testimony and other evidence has been received; any allowances for post-hearing briefs before the matter is deemed submitted to the hearing officer do not extend this time.

7.15. Transcript of Hearing

- a. The Association shall order a transcript of the hearing for the hearing officer's use. The costs shall be paid by the Association.
- b. If a copy of the transcript is ordered at the request of the applicant or the applicant's attorney, the costs shall be paid by the applicant.

c. The costs of a copy ordered by any party after the original transcript has been prepared shall be borne by the party ordering the copy.

7.16. Post-Hearing Briefs

Not later than thirty (30) days after the conclusion of the hearing, the parties may file simultaneous written briefs with the Association. The hearing officer may extend this deadline for good cause to 45 days. Alternatively, the parties may stipulate to, or the hearing officer may require, opening, opposing, and reply briefs, on a schedule to be determined by the parties, with the written description of the schedule to be provided promptly to the Association.

Any party which wishes to file a written brief shall file an original with the Association, with a proof of service on the hearing officer and the other party or parties, which sets forth the issues and contentions of the party and a summary of the evidence, and discusses applicable law. (For help in writing a brief, parties may search the Internet for "how to write a legal argument in a court case.") References to factual matters contained in the record must include a citation to the volume and page of the exhibits where this information is shown.

One original of each brief must be filed in hard copy with the Association. The parties may stipulate to serve solely electronic copies on each other, and the hearing officer may consent to be served solely with an electronic copy.

The matter shall be deemed submitted on the day the post-hearing briefs are, or the reply brief is, due. If the parties waive the filing of post-hearing briefs, the matter shall be deemed submitted at the conclusion of the hearing.

7.17. Failure to Pursue Application

If, as a result of the applicant's failure to comply with the procedures specified above, **including but not limited to failure to cooperate in scheduling the hearing, or unwarranted requests to postpone or continue a hearing,** the hearing is not concluded within nine (9) months after a request or referral for a formal hearing, the case shall be dismissed with prejudice. This remedy pertains to a matter that has already been sent to a hearing and does not supersede the dismissal without prejudice provided for in section 2.5 (time limit for filing application).

7.18. Findings of Fact and Recommendations

a. Not later than thirty (30) days after the date the matter is deemed submitted, as provided in section 7.16, the hearing officer shall serve his/her proposed Findings of Fact and Recommendations on the parties and file them with the Board. The hearing officer may file with the Association a written request for one 30-day extension for submission of the recommended decision, which request shall be granted or denied by the Retirement Administrator.

One original of the proposed Findings of Fact and Recommendations must be filed in hard copy with the Association. The parties may stipulate to allow the hearing officer to serve solely an electronic copy on them.

The hearing officer shall include in his/her recommended decision a notice to the parties that they have ten (10) days from the date of service of the recommended decision to submit written objections, with a proof of service on the other parties, to the Board.

b. Pursuant to Government Code section 31534, the parties shall have ten (10) days from the date of service of the hearing officer's proposed Findings of Fact and Recommendations to file written objections, with a proof of service on the other party or parties, with the Association. (The objections are **not** served on the hearing officer.) These objections shall be incorporated in the record to be considered by the Board when it reviews the proposed decision.

One original of any objections must be filed in hard copy with the Association. The party receiving any objections may consent to be served solely with an electronic copy.

7.19. Shortening Time Periods

Nothing in these Procedures is to be construed as preventing the parties from stipulating to lesser intervals than those prescribed above. The hearing officer may also, for good cause shown, after giving both parties an opportunity to be heard, shorten the times specified in this Article. (See also section 5.5 regarding hearing officer's authority to waive or modify some procedures.)

8. BOARD DECISION AFTER HEARING

8.1. Board's Options

Upon receipt of the hearing officer's proposed Findings of Fact and Recommendations and considering the record, including any objections submitted by the parties, the Board may, in accordance with Government Code section 31534:

- a. Approve and adopt the proposed findings and the recommendations of the hearing officer;
- b. Require a transcript, plus all other evidence received by the hearing officer and, upon the receipt thereof, take such action as in its opinion is indicated by such evidence;
- c. Refer the matter back with or without instructions to the hearing officer for further proceedings; or

d. Set the matter for hearing before itself. At such hearing the Board shall hear and decide the matter as if it had not been referred to the hearing officer.

8.2. Referral Back to Hearing Officer

If the Board refers the matter back to the hearing officer with a request to amend the proposed decision, the hearing officer shall submit the revised proposed decision within 30 days after the date of the request from the Board.

If the Board refers the matter back to the hearing officer with instructions for further proceedings, the Board shall include in its instructions a deadline for completion of the additional proceedings.

8.3. Findings and Conclusions by the Board

In any case where the Board sets the matter for hearing before itself, the Board shall adopt such findings of fact and conclusions of law as it deems appropriate.

8.4. Notice of Decision

The Association shall give written notice of the Board's decision to all parties and their representatives within ten (10) days. Said notice shall include the decision and the date it was rendered.

8.5. Finality of Decision

Except as otherwise provided in Article 10 (judicial review), action by the Board granting or denying in whole or in part any application for disability retirement shall be deemed to be a final and conclusive determination of any and all issues raised by the application which were either presented to the Board or could with the exercise of reasonable diligence have been presented to the Board. No employee whose application for disability retirement has been denied by the Board, and who thereafter returns to service with a participating employer, shall be precluded from filing a new application which asserts as a basis for the application with circumstances which have arisen or become known after denial of the first application. Such new application shall not, however, relate back to the date of an original application that was denied, and any entitlement to benefits shall be effective as of the date of filing of the new application.

PART IV.

MISCELLANEOUS

9. CHANGE OF POSITION IN LIEU OF DISABILITY RETIREMENT ALLOWANCE

As required by Government Code sections 31725.5 and/or 31725.65, if a disability applicant is found disabled and returns to County or other participating employer service in a position which pays less than the position from which he/she became disabled, the Association shall pay, in lieu of a disability retirement allowance, the difference between the member's former salary and current salary. Members should also note that if a safety member is placed in a general position, the member retains safety status.

10. JUDICIAL REVIEW

10.1. Judicial Review

In those cases where a party is entitled to judicial review of the proceedings, the petition to the court shall be filed in accordance with Code of Civil Procedure section 1094.5. Notice of the right to review under section 1094.5, and the time limits for filing for such review under Code of Civil Procedure section 1094.6, shall accompany the written notice of the Board's decision.

If the member's application was denied on the ground that the medical examination and other available information did not show to the satisfaction of the Board that the member was incapacitated physically or mentally for the performance of his/her duties, the notice to the employer shall advise the employer that the employer's time limit for obtaining judicial review is governed by Government Code section 31725, which requires the employer to obtain judicial review of the Board's action by filing a petition for writ of mandate within 30 days after the mailing of the notice. If the employer's application was denied on this ground, the notice shall advise the employer that this deadline **may** apply.

10.2. Preparation of Administrative Record

Pursuant to Code of Civil Procedure section 1094.6, TCERA staff shall prepare the administrative record for court upon written request by the petitioner. The petitioner must pay the cost of preparing the original of the administrative record in hard copy for the court. The petitioner must pay the estimated cost of preparation before TCERA prepares the administrative record. If the litigation is abandoned or dismissed before TCERA has completed preparation of the administrative record, or if the deposit exceeds the preparation cost, TCERA will refund any portion of the deposit exceeding the preparation cost actually incurred.

Pursuant to Code of Civil Procedure section 1094.5, the petitioner may ask the court to find that the petitioner cannot afford to pay any court costs, including the cost of the administrative record. If the court issues an order finding the petitioner to be indigent, and the petitioner provides a copy

of the signed order to TCERA, TCERA shall then pay the cost of providing the original of the administrative record in hard copy for the court. (This cost is recoverable as court costs if TCERA prevails in the litigation.) TCERA shall file the original with the court.

TCERA will provide a hard copy of the administrative record to its counsel at no charge. Any party other than TCERA must pay a fee to obtain a copy, and may choose what type of copy to receive: only an electronic copy (minimal fee), only a hard copy (higher fee), or both. If the petitioner obtains a court order of indigency, the procedure set forth above for the original of the administrative record (which is submitted to the court) shall also apply to the copy for the petitioner.

11. RE-EXAMINATIONS

The Board may, pursuant to Government Code section 31729, require any disability retiree under age 55 to undergo medical examination. The examination shall be made by a physician appointed by the Board. Upon the basis of the examination the Board shall determine whether the disability retiree is still physically or mentally incapacitated for service in the job classification held by him/her when retired for disability. The Board may additionally, or in the alternative, arrange for an investigation of the circumstances pertaining to such retiree that relate to his/her entitlement to receive disability benefits.

In the event that the examination or investigation shows that the retiree may not be incapacitated, the Board shall thereupon notify the retiree, at his/her last known address, of said recommendation of the medical advisor and advise the retiree that the Board will meet to reconsider his/her disability retirement status. If the Board's decision is to refer the matter to a hearing officer for a determination, the rules as set forth in Articles 5 through 7 (prehearing procedures, hearing on the record procedures, and formal hearing procedures), to the extent applicable, shall apply. Said notice shall inform the retiree of his/her right to request a formal hearing, provided said request is submitted in writing no later than fifteen (15) days after the date of the notice.

12. SUBROGATION

Disability applicants are hereby notified that the 1937 Act provides for subrogation of benefits. (See Gov. Code, § 31820 et seq.) This means that if the Association is paying benefits in a situation where someone other than the employer is responsible for the injury, the Association may seek to obtain partial reimbursement from that person of the benefits paid. In doing so, the Association may join with the employer or the employer's workers' compensation carrier in any proceeding, or may file its own action or a claim in an action filed by the employee.

PART V.

EMPLOYER-FILED APPLICATIONS

13. EMPLOYER-FILED APPLICATION

13.1. Applicability

The provisions of this Article shall be in accordance with Government Code section 31721. This statute provides that an employer may not separate because of disability a member otherwise eligible to retire for disability, but shall apply for disability retirement of any eligible member believed to be disabled. The Association shall provide appropriate information to the employer regarding eligibility of a member for disability retirement upon request by the employer.

All provisions of these Procedures shall be applicable, substituting the employer-applicant for the member-applicant, except as provided below.

13.2. Application

The application for disability retirement shall consist of each and all of the following:

- a. A completed Application for Disability Retirement;
- b. A written statement signed by the department head or his/her authorized agent which asserts that a member of the Association is disabled in a manner which permanently incapacitates the member from the substantial performance of his/her duties, describes the nature and extent of the disability, and identifies the effect which the disability has had upon performance by the member of his/her duties; and
- c. A written statement which is as close as the employer can come to the equivalent of the treating physician's statement, which indicates that the member is permanently incapacitated for the performance of his/her duties.
 - If the employer has independent subpoena power (such as the Board of Supervisors' power under Gov. Code, § 25170 et seq.), and the employer is unable to obtain a written statement from the member's treating physician, this statement shall consist of, at a minimum, a written opinion from a physician who has reviewed the medical records obtained by the employer pursuant to its subpoena power.
 - 2) If the employer does not have independent subpoena power, and the employer is unable to obtain a written statement from the member's treating physician, the Association may accept the application without complete information, provided the written statement by the employer sufficiently

shows why the employer believes the member is permanently incapacitated, or may reject the application until the employer provides more information supporting its belief that the member is permanently incapacitated. If the Association accepts the application, the Association shall then (notwithstanding section 7.4 prohibiting discovery during the hearing stage) provide to the employer such deposition and other_subpoenas as requested by the employer during the investigation stage, so that the employer may have medical records sent to TCERA, and may then make arrangements with TCERA to have a physician review the medical records and provide a written opinion based on the records to the Association regarding whether the member is permanently incapacitated, and if yes whether on a serviceconnected basis or not.

d. The application will be assumed to be on a non-service basis unless the employer affirmatively states that the employer believes the incapacity is service-connected. As set forth in section 13.3.d., the member may later assert that the incapacity is service-connected.

13.3. Notice

The Association shall, not later than ten (10) days after the date of the application is filed, personally, or by certified mail, serve on the member the following:

- a. A copy of these Procedures.
- b. A copy of the employer's application.
- c. The Disability Retirement Questionnaire and the release forms to enable the Association to obtain the Applicant's medical and personnel records.
- d. A written notice which informs the member:
 - 1) Of the application and the date of its filing;
 - 2) Of the member's right to either support or oppose the application and to be represented by legal counsel at the member's expense;
 - 3) Of the member's right to provide one or more medical reports pertaining to the application to the Association, at the member's expense.
 - 4) Of the member's right to amend the application to become the applicant, and that, if the member does so, the member may also amend the application for non-service-connected disability retirement to apply for serviceconnected disability retirement, or that, alternatively, the member may file a new application for service-connected retirement after the employer has shown that the member is permanently incapacitated.

- 5) Of the requirement that, if the member becomes the applicant, the member shall provide all information required as described in Article 2.
- 6) That, unless the member becomes the applicant, the employer shall have sole control of any decisions normally made by the applicant.
- 7) That any communications between the Association and the member shall not be deemed to substitute for any communications between the employer and the member.

13.4. Burden of Proof

Where the employer has filed the application, the employer has the burden of showing permanent incapacity. However, as provided in section 13.3.d.4., the member may either amend an employer-filed application to become the applicant, or wait and file a new application. The burden of proof and the procedures shall therefore be as follows.

- a. Unless and until the member amends the application to become the applicant, the employer shall have the burden of proof to show permanent incapacity and, if applicable, service-connectedness.
- b. If the member amends the application to become the applicant, the member shall have the burden of proof to show permanent incapacity and, if applicable, service-connectedness.
- c. The member's right to amend the application to become the applicant shall expire when the Board makes its initial decision as described in section 13.7 below. After that date, the member must file a new application for any type of disability not granted by the Board in the initial decision, within the time provided in this Article.

13.5. Costs

Even if the employer is the applicant, all physician, medical, and other costs incurred for any and all examinations and medical reports undertaken or prepared **at the request of the member** shall be borne solely by the member. The employer and member may agree that the costs of any examination or medical report will be paid by the employer.

13.6. Review of Application

Review of an employer-filed application shall be in the same manner as a member-filed application, except that where the member is not cooperating in the review process, the Association will not schedule a medical examination to be paid for by the Association.

13.7. Initial Decision of Board Where Member Has Not Amended Employer's Application to Become Applicant

The initial decision by the Board shall be in accordance with Article 4, except that the actions described below shall result in the following procedures:

a. Decision to grant employer's application for non-service connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board. The notice to the member shall include information regarding the member's right to request a hearing in the event the employee is in disagreement with the decision of the Board. The employee must file the request for hearing within 10 days from the date of the notice, in the same manner as where the member is the applicant, except that the hearing shall not consider the issue of service connection.
- 2) At the close of the 10-day period for the member to request a hearing, the Association shall notify the employer whether the member has filed a request for hearing regarding the decision of the Board.
- 3) If the member does not file a request for hearing, the Board's decision as to permanent incapacity shall be final. The member shall then be sent a notice stating that the member has 10 days from the date of the notice to file a **new** application for service-connected disability retirement.
- 4) At the close of this second 10-day period, the Association shall notify the employer whether the member has filed a new application for serviceconnected disability retirement. If no new application is filed, the Association shall notify the parties that the matter is final.

b. Decision to <u>deny</u> employer's application for non-service-connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board.
- 2) The notice shall include information regarding the employer's right to request a hearing in the event that the employer is in disagreement with the decision of the Board. The notice to the employer shall include the information that failure to request a hearing shall result in the member being reinstated to his/her employment, in accordance with Government Code section 31725. The employer must file the request for hearing within 10 days after the date of the notice, in the same manner as where the member is the applicant, except that the hearing shall not consider the issue of service connection. A member who has not amended the application to become the applicant may not request a hearing.

- 3) If the employer files a request for hearing, the matter shall proceed as described in Articles 5-7.
- 4) If the employer does not request a hearing, the Board's decision shall be final, and the Association shall so notify the employer and member.
- 5) If the employer originally requests a hearing, and then later requests to withdraw the application before the hearing, the hearing will be vacated and the decision denying the application shall be final. (A **new** application may be filed thereafter by any party.)

c. Decision to grant employer's application for service-connected disability retirement

- The parties shall be notified within 10 days of the decision of the Board. The notice to the member shall include information regarding the member's right to request a hearing in the event the member is in disagreement with the decision of the Board. The employee must file the request for hearing within 10 days after the date of the notice, in the same manner as where the member is the applicant.
- 2) At the close of the 10-day period for the member to request a hearing, the Association shall notify the employer whether the member has requested a hearing regarding the decision of the Board.
- 3) If the member does not file a request for hearing, the Board's decision shall be final, and the Association shall so notify the parties.

d. Decision to <u>grant</u> non-service-connected disability retirement, but to deny employer's application as to service connection

- The employer and member shall be notified within 10 days of the decision of the Board. The notice shall inform the employer that the employer may request a hearing on the question of service connection, and shall inform the member that the member may request a hearing on the question of permanent incapacity. Either request must be filed within 10 days from the date of the notice.
- 2) If any party files a request for hearing, the matter shall proceed as described in Articles 5-7, except that if the member is the only party who requests a hearing, the hearing shall not include the issue of service connection. If one party requests a formal hearing and the other party requests a hearing on the record, the procedures for a formal hearing shall apply.

- 3) If no party requests a hearing, the Board's decision shall be final, and the Association shall so notify the parties.
- 4) If the employer requests a hearing, but later withdraws the request for hearing, the member shall then be sent a notice stating that the member's request for hearing shall continue, in the form of hearing requested by the member, but only on the question of permanent incapacity, and that if the member wishes to reinstate the issue of service connection, the member has 10 days to file a **new** application for service-connected disability retirement.

e. Decision to <u>deny</u> employer's application for service-connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board. The notice shall include information regarding the employer's right to request a hearing in the event that the employer is in disagreement with the decision of the Board. The notice to the employer shall include the information that failure to request a hearing shall result in the member being reinstated to his/her employment effective as of the day following the effective date of dismissal, in accordance with Government Code section 31725. The employer must file a request for hearing within 10 days from the date of the notice.
- 2) If the employer requests a hearing, the matter shall proceed as described in Articles 5-7.
- 3) If the employer does not request a hearing, the Board's decision shall be final.
- 4) If the employer originally requests a hearing, and then later withdraws the request for hearing, the hearing will be vacated, and the decision denying the application shall be final. (A new application may be filed thereafter by any party, with a new effective date if granted.)

f. Decision to dismiss application without prejudice

- 1) The employer and member shall be notified within10 days of the decision of the Board. The notice shall provide information regarding the employer's and member's rights to re-apply for a disability retirement.
- 2) Should the employer or member re-apply for a disability retirement, the effective date of disability retirement, if granted, will relate to the date of the re-application as required by statute.

13.8. Initial Decision of Board Where Member Has Amended Employer's Application to Become Applicant

Where the employer has applied for disability retirement, but the member has amended the application to become the applicant, with or without adding a request for service-connected disability retirement, if applicable, the initial decision of the Board shall be in accordance with Article 4.

13.9. Hearing Procedure

The hearing procedure shall be the same for an employer-filed application as for an application filed by a member, except that the member and his/her legal counsel shall also have the opportunity to participate, and shall have all the same rights and duties as the employer-applicant and its legal counsel and the Association and its legal counsel.

The hearing officer may make any rulings and/or modify the hearing process, except for the time limits provided in section 7.3, as needed to provide a fair hearing involving a third party.

13.10. Board Decision after Hearing

Where the employer remains the applicant, the decision of the Board after a hearing shall generally be made as provided in Article 8, with "employer" substituted for "member" where appropriate.

However, where the employer applied only for non-service-connected disability retirement, and the Board approves a recommendation by the hearing officer to grant the application, the initial notice of the Board's decision under section 8.3 shall not state that the decision of the Board is final. Instead, the notice shall state that the member has 10 days to file a **new** application for service-connected disability retirement.

If no new application for service-connected disability retirement is filed, at the close of this 10-day period the Association shall notify the parties that the matter is final.

If a new application for service-connected disability retirement is filed, the Association shall inform the employer of this fact, and shall advise the parties that the matter will not be final until the Board reaches a decision on the question of service connection.

The new application for service-connected disability retirement shall be deemed timely if the employer's application was filed within the time limits in Government Code section 31722, as described in section 2.5 of these Procedures. The new application shall be processed and decided upon as provided in these Procedures, except that the additional materials normally required to be included with an application by a member need not be provided at the same time, but instead may be provided within 30 days after the application is filed.

13.11. Judicial Review

The opportunity to obtain judicial review shall be as set forth in Article 9.

BBG/20171017/3/26/20/

APPENDIX A

STATE OF CALIFORNIA) COUNTY OF TULARE

I, _____, hereby declare that I am the applicant (attorney for the applicant) (attorney for the Association regarding the application of ______) for a disability retirement pension. I believe that I (the applicant)

(the Association) cannot have a fair and impartial hearing before the hearing officer to whom the case is assigned.

I declare under penalty of perjury that the foregoing is true and correct.

Executed on this_____ day of ______, 20___, at _____, California.

Signature

TULARE COUNTY EMPLOYEES' RETIREMENT ASSOCIATION (TCERA)

DISABILITY RETIREMENT PROCEDURES

Approved by the Board of Trustees on September 28, 2016 Amendments: May 10, 2017, May 13, 2020, April 13, 2022

TULARE COUNTY EMPLOYEES' RETIREMENT ASSOCIATION (TCERA) DISABILITY RETIREMENT PROCEDURES (Effective May 13, 2020)

PART I. GENERAL PROVISIONS

1. GENERAL PROVISIONS

1.1. Purpose

The purpose of these Procedures is to provide an equitable, fair and impartial method for acting upon applications for rights, benefits and privileges under the County Employees Retirement Law of 1937 (Gov. Code, § 31450 et seq.), to the end that applications may be expeditiously processed.

1.2. Definitions

In these Procedures, unless the context or subject matter otherwise requires:

- a. "Accommodations" means an agreement by the County or other participating employer to modify the regularly assigned duties and/or work environment of the member as necessary to allow the member to perform his/her regularly assigned duties, as modified, within his/her current job classification and within the limitations established by medical evidence.
- b. "Administrative record" means the record prepared by TCERA staff for proceedings in court, pursuant to Code of Civil Procedure section 1094.6.
- c. "Applicant" means: (1) a member of TCERA claiming benefits, rights, or privileges under the County Employees Retirement Law of 1937, (2) any person claiming such benefits, rights or privileges through a member, or (3) the County of Tulare or other participating employer in TCERA.
- d. "Association" or "TCERA" means the Tulare County Employees' Retirement Association.
- e. "Board" means the Board of Retirement.
- f. "Brief" means a written legal argument explaining why that side should win the case. References to factual matters contained in the record must include a citation to the volume and page of the exhibits where this information is shown.

- g. "County health officer" means the county health officer appointed by the Board of Supervisors pursuant to Health and Safety Code section 101000 et seq. (Pursuant to Gov. Code, § 24100, this term also includes any deputy.)
- h. "Day" means calendar day. The time to take an action specified under these Procedures shall be governed by general California law (see Code Civ. Proc., § 10 et seq.). For example, if the last day to file a document with the Association is a date when the Association office is closed prior to the end of regular business hours, or for the entire day (on a weekend or holiday or for some other reason), then the time period is extended to the next business day.
- i. "Disability" or "disabled" means permanent physical and/or psychological inability to substantially perform the regularly assigned duties of the job, with or without accommodations.
- j. "Effective date" of a disability retirement allowance means the date calculated pursuant to Government Code section 31724. As of the last date of amendment of these Procedures, this statute provides that the effective date of a disability retirement allowance can be:
 - 1) The date the disability retirement application is filed with the Board;
 - 2) Not earlier than the day following the last day for which the member received regular compensation;
 - 3) After the expiration of any sick leave with compensation, unless the member agrees to retirement at an earlier date; or
 - 4) After the expiration of any leave of absence with compensation paid by the employer pursuant to Labor Code section 3201 et seq., unless the member agrees to retirement at an earlier date.

The statute also provides that when it has been demonstrated to the satisfaction of the Board that the filing of the member's application was delayed by administrative oversight or by inability to ascertain the permanency of the member's incapacity until after the date following the day for which the member last received regular compensation, such date will be deemed to be the date the application was filed.

- k. "Filed" means actually received by TCERA during regular business hours.
- *l*. "Formal hearing" means a hearing with live testimony by sworn witnesses, in addition to the hearing exhibits. Compare "hearing on the record."
- m. "Health care provider" means any person or entity which has provided medical or mental health care or treatment and has records which may be relevant to the disability retirement application, including but not limited to a physician, dentist,

psychologist, chiropractor, optometrist, physical therapist, acupuncturist, counselor, hospital, medical clinic, or laboratory.

- n. "Hearing exhibits" means the documents prepared by TCERA, plus those submitted by the parties, for a hearing.
- o. "Hearing officer" means an active member of the State Bar of California appointed by the Board pursuant to Government Code section 31533 to independently review the evidence and provide proposed findings of fact and recommendations to the Board, pursuant to the provisions of law and these Procedures.
- p. "Hearing on the record" means a procedure in which the hearing officer makes his/her recommendation based only on the hearing exhibits and any written legal arguments submitted by the parties (i.e., without live testimony). Compare "formal hearing."
- q. "Legal counsel" means an attorney who is an active member of the State Bar of California.
- r. "Member" means a member of TCERA.
- s. "Party" means any person disclosed by the records of the Association or by the application to have an interest in the subject matter of an application. Where appropriate within the context of the provision, this term may be limited to mean only the official participants in a hearing.
- t. "Pending" means the period from the time an application has been accepted as complete and ready for processing, through the final decision of the Board.
- u. "Permanent incapacity for performance of duty" means a permanent disability that precludes a member from substantially performing the regularly assigned duties of his/her job classification, even with accommodations.
- v. "Physician" means:
 - 1) A person holding a valid and unrevoked physician's and surgeon's certificate or certificate to practice medicine and surgery, issued by the Medical Board of California or the Osteopathic Medical Board of California; or
 - 2) A person with a doctoral degree in psychology, to whom a license has been issued pursuant to the provisions of California law, which license is in force and has not been suspended or revoked.
- w. "Retirement Administrator" means the individual employed as the Retirement Administrator by TCERA. The term includes a designee acting as Retirement

Administrator in the absence of the incumbent Administrator, as well as an Acting or Interim Retirement Administrator if the position of Retirement Administrator is vacant.

x. "Service" in certain contexts in these Procedures means that period of time for which deductions are made from a member's earnable compensation from the county or district. The "last day of service" is the last day of the last pay period for which retirement contributions are deducted. A member is considered to have discontinued service after this date.

For a member in a tier in which contributions end after thirty years of retirement service credit (Tiers 1-3), the "last day of service" may instead be the last day of the last pay period for which retirement contributions would have been deducted if the member did not have thirty years of retirement service credit. A member is considered to have discontinued service after this date.

- y. "Service" in other contexts in these Procedures means formally sending a document under these Procedures. "Proof of service" means a document certifying that a related document has been properly served on the required recipient(s) under these Procedures. Proper service means delivery of a document by one of the following methods:
 - 1) In person.
 - 2) Depositing the document in first-class mail, postage pre-paid, addressed to the recipient at the address last known to the sender. Delivery is deemed complete when the document is mailed.
 - 3) Between TCERA and an employer which participates in the County interoffice mail system, by interoffice mail. Delivery is deemed complete when the document is picked up by the courier.
 - 4) With the written consent of the recipient, by email, facsimile, or other electronic means. Delivery is deemed complete when the document is sent, provided the sender does not receive notice that the document was not sent. (Note: TCERA has consented in these Procedures to receive electronic notice from the employer regarding counsel for the employer. See section 1.6.a.)
 - 5) Any document that is sent electronically after the close of business on that day shall be deemed to have been sent on the next business day.
- z. "Subrogation" means TCERA's right to recover retirement benefits paid to a member from a third party who caused the permanent incapacity of the member.

1.3. Burden of Proof

a. In an application for non-service connected disability retirement, the burden rests with the applicant to prove that he/she is permanently disabled from substantially performing the regularly assigned duties of his/her job.

Whenever an issue of drug or alcohol abuse is presented by the evidence, the burden set forth in this subdivision shall include the burden to prove that the particular incapacity was not the result of the alcohol or drug abuse.

- b. In an application for service-connected disability retirement, the burden rests with the applicant to prove that he/she is permanently disabled from substantially performing the regularly assigned duties of his/her job **and** that the employment contributed substantially to the disability.
- c. See Article 13 for the burden of proof rules where the employer is the applicant.

1.4. Communication with Individual Board Members

The Board acts as judge and jury regarding all disability retirement applications. As such, once an application for disability benefits has been filed, all communications between individual Board members and interested parties or their representatives regarding the application are forbidden until such time as the Board issues its written decision. Applicants are advised that any violation of this rule may result in the Board member being found ineligible to participate in any discussion regarding the application, or to influence the Board's decision, or to vote on the disability application.

- a. If a Board member receives a communication in violation of this provision, the Board member shall immediately take one of the following steps:
 - 1) If the communication is written, the Board member shall immediately provide to the Retirement Administrator a copy of the writing and any written response by the Board member to the communication.
 - 2) If the communication is oral, the Board member shall immediately send an email to the Retirement Administrator, stating the substance of the communication, any response made by the Board member, and the identity of each person from whom the Board member received the communication.
- b. The Retirement Administrator shall include the information provided by the Board member in the materials presented to the Board, and in the record. The Retirement Administrator shall notify the parties that the communication has been made part of the record.
- c. If a party requests an opportunity to address the Board regarding the communication within 10 days after the date of the notice from the Retirement Administrator, the Board may allow the party to do so.

1.5. Requirements for Board Member Participation

Any board member who has not reviewed the record that was provided to the entire Board for consideration shall not vote.

1.6. Right to Counsel

a. An applicant is entitled, at his/her expense, to be represented by legal counsel (as defined in section 1.2.q). An applicant may not be represented by a person who does not meet this definition.

Where the member is the applicant, the Association will recognize legal counsel as the representative of an applicant only upon receipt of a document signed by the applicant which lists the contact information for the attorney, including the attorney's California state bar number. Where the County is the applicant, the Association will recognize the appropriate team within the County Counsel's office as the representative of the applicant without documentation, or will recognize another legal counsel as the representative of an employer applicant upon receipt of any form of written notice from the employer (including an email) which lists the contact information for the attorney, including the attorney's California state bar number.

The Association reserves the right to contact the applicant directly in addition to or in lieu of contacting the applicant's legal counsel.

b. The Association is entitled to be represented by legal counsel, and shall be deemed to be represented by the appropriate team within the County Counsel's Office, unless otherwise designated by the Board.

1.7. Effect of Stipulations by Employer

The Board is not bound by any stipulation, agreement, understanding, act, or omission on behalf of the County of Tulare or other participating employer in TCERA by an officer, employee or agent of the County of Tulare or other participating employer in TCERA, in a workers' compensation or any other proceeding to which neither this Board nor the Association is a party, insofar as such stipulation, agreement, understanding, act, or omission may be asserted to relate to disability retirement or the disposition of a particular application for disability retirement.

1.8. Waiver of Procedures

a. The Board may waive or modify any provision of these Procedures, in whole or in part, to ensure the fairness of these Procedures for all parties, including the Association.

b. During an emergency which has been officially declared by an appropriate state official (as provided in Gov. Code, § 8680 et seq.), or by an appropriate County official (as provided in Tulare Co. Ord. Code, § 1-15-1000 et seq.), any party may request, and the Retirement Administrator may approve, an extension of any deadline in these Procedures until a reasonable period of time after the next scheduled meeting of the Board. If the Board is unable to take action at the next meeting as provided in section 1.8.a. for any reason (for example, losing a quorum or having a tie vote), or if the next meeting cannot be held until after the extension expires, the Retirement Administrator may continue to approve requests for extension until the Board is able to act. Both the requests and the approvals shall be in writing (email is acceptable).

1.9. Denial Due to Fraud or Willful Misrepresentation

If the applicant engages in any of the following actions, the application shall be denied with prejudice. Actions leading to denial on this ground include, but are not limited to:

- a. Willfully misrepresenting any fact or omitting relevant information in order to obtain disability retirement benefits.
- b. Presenting or causing to be presented any knowingly false or fraudulent written or oral statement in support of an application for disability retirement. For the purposes of this provision, the term "statement" includes records submitted to support the application.
- 1.10. Dismissal Due to Non-Compliance or Failure to Cooperate

In any matter in which the Association determines that the applicant has failed to comply with these Procedures, or to cooperate with the Association as reasonably required for the Association to process the application, the Association may dismiss the application without prejudice. The matter will be placed on a Board of Retirement meeting agenda for discussion and action regarding the dismissal. For purposes of this subsection, to "dismiss the application without prejudice" shall mean that the applicant may re-apply for a disability retirement whether or not the new application will be based upon the same facts and circumstances. If a new application is filed, the effective date of any disability retirement, if granted, will relate to the date of the new application as required by statue, and not to the date of the application which was dismissed.

1.11. Withdrawal

- a. An applicant may withdraw an application at any time prior to the initial decision of the Board.
- b. An applicant may withdraw a request for hearing at any time between the initial decision of the Board and the final decision of the Board.
 - 1) The withdrawal must be in writing and must be filed with the Association.

2) The withdrawal is irrevocable once filed, and makes the initial decision of the Board the final decision. The decision cannot be appealed to court because without a hearing the applicant has not exhausted administrative remedies.

1.12. No Additional Time for Mailed Documents

The extension of time provided by Code of Civil Procedure section 1013 when a document is served by mail does **not** apply to any deadlines set by these Procedures.

PART II. APPLICATION THROUGH INITIAL BOARD DECISION

2. APPLICATION FILED BY MEMBER¹

2.1. Furnishing Copy of Procedures

The Association shall include a copy of these Procedures in the packet provided to anyone wishing to file an application. The Association shall also provide a copy to any member of the public on request, or, if these Procedures are posted on the TCERA website, may direct the requesting person to the website.

2.2. Filing Application

The Association shall furnish application forms for disability retirement benefits upon request to any person authorized to file an application. The applicant shall fully complete the application forms in accordance with these Procedures and the instructions stated on the application forms, and shall sign and verify the completed forms under penalty of perjury.

The completed application shall be filed with the Association. Any application or document filed in support thereof that is incomplete or illegible shall not be accepted for filing, but instead shall be returned to the applicant.

2.3. Documents to Be Included in Application

The following documents and reports shall be completed or furnished by the applicant in connection with the initial application. All documents must be legible and capable of being photocopied legibly.

- a. Completed Application for Disability Retirement.
- b. Responses to Disability Retirement Questionnaire, signed under penalty of perjury. (See section 1.9 regarding misrepresentation or omission of relevant information.)

¹ See Article 13 for provisions regarding employer-filed applications.

The applicant shall list **all** health care providers seen by the applicant beginning with the date which is five (5) years prior to the date of the injury, diagnosis, or other disabling condition, and continuing through the date of application. The applicant shall also list all providers seen earlier than this five (5)-year date regarding the same or a similar injury, diagnosis, or condition, and shall include the approximate dates seen.

The applicant shall sign and attach to his/her responses the signature page which the Association provides with the Questionnaire which includes the declaration under penalty of perjury. If the responses are not returned with the appropriate signature page they shall be deemed not to have been filed and shall be returned to the applicant with additional instructions.

- c. Completed release forms to enable the Association to obtain the Applicant's medical and personnel records.
- d. Completed Treating Physician Statement(s).

The Treating Physician Statement (TPS) must state that the member is permanently incapacitated for the performance of the duties of his/her position, and, if the application is for service-connected disability retirement, that the member's employment contributed substantially to such incapacity. The TPS must be signed by a_physician (as defined in these Procedures) with expertise in the type of disability claimed by the member.

If the TPS does not state that the member's employment contributed substantially to the incapacity, the application will be treated as one for non-service-connected disability retirement, even if the applicant has stated in the application that he/she believes the disability is directly related to employment with an employer participating in TCERA.

e. Additional and supplemental Releases and Questionnaires may be required in the course of review of the application.

2.4. Basis for Application

A disability retirement application shall be based upon a permanent medical condition which constitutes a physical or psychological incapacity, or a combination of both. The documents on file must clearly identify the exact nature of the incapacity that forms the basis for the disability application. A disability retirement will not be granted except for those incapacities stated in the application on file and proven by the applicant to the satisfaction of the Board.

2.5. Time Limit for Filing

- a. An application for disability retirement shall not be deemed effective or filed within the meaning of Government Code sections 31721 through 31724, inclusive, until the date on which it is received by the Association at the office indicated in Section 2.6. An amended application shall be deemed filed, within the meaning of these statutes, on the date it is received by the Association. It is the applicant's responsibility to obtain such documents and the Association shall not process any disability retirement application until all the information required in Section 2.3 has been furnished, at which time the application shall be deemed complete and ready for processing.
- b. The burden rests with the applicant to establish timeliness of filing. As required by Government Code section 31722, the application must be filed:
 - 1) While a member is in service;
 - 2) Within four months after the member's discontinuance of service;
 - 3) Within four months after the expiration of any period during which a presumption is extended beyond the member's discontinuance of service; or
 - 4) While, from the date of discontinuance of service to the time of the application, the member is continuously physically or mentally incapacitated to perform his/her duties.

Applicants should note that if the application is filed under this ground, the applicant has an additional burden regarding timeliness (proving continuous disability) which does not apply to the first three grounds.

- c. See definition of "service" in section 1.2.x.
- d. The applicant may show that the member is entitled to an effective date of disability retirement earlier than the date of the application, pursuant to Government Code section 31724. (See definition of "effective date" in section 1.2.j.)

2.6. Where to File

Application forms and related materials such as these Procedures may be obtained from the Association, 136 North Akers Street, Visalia, California 93291. All forms and documents shall be filed in hard copy at this location unless otherwise specified. All telephone inquiries should be made to (559) 713-2900.

2.7. Amendment of Application

- a. An applicant may amend his/her application at any time before the Board's initial decision. The new information shall be added to the staff review and recommendation to the Board.
- b. After the Board's initial decision, an application may not be amended. The applicant must instead file a new application, which will be held without investigation until the pending application has been finally decided, and may then file a new application. The effective date of benefits pursuant to the new application shall be determined as if no prior application had been filed.

3. STAFF REVIEW AND RECOMMENDATION

3.1. Review of Application

Once all required information has been submitted and the application is deemed complete and ready for processing, the Association shall process the application as set forth below.

3.2. Records in Support of Claim

- a. The applicant shall make arrangements for **all** medical records from the health care providers listed in the responses to the questionnaire to be sent directly to the Association by the physician, medical clinic, hospital, or other health care provider. Records submitted directly by the applicant will be rejected, unless the health care provider seals the records in a manner acceptable to the Association.
- b. The applicant may submit a written request to the Association to be excused from providing particular records, for good cause. The request shall be signed under penalty of perjury. The Association in its sole discretion may grant or deny the request, and the response shall also be in writing. The Association may provide a form for the request.
- c. The Association at any time may demand that the applicant arrange for additional records which the Association believes may be relevant to the application to be provided to the Association. Disputes regarding such records may be referred to the Retirement Administrator. The Association may also require the Applicant to provide additional and supplemental releases and responses to questions in the course of review of the application.

3.3. Medical Examinations

a. A member may be required to submit to medical examinations by a physician or other expert identified by the Association. The Association shall pay for the cost of any such examination. Refusal of any applicant to submit to such medical examination shall be grounds for considering that the results of such examination would be unfavorable to the applicant, and such refusal shall result in denying the application with prejudice. The cost for an Independent Medical Examination or review estimated to exceed the amount set from time to time by the Board shall be approved in advance by the Board.

- b. The Association will reimburse vehicle mileage to and from a medical examination outside Tulare County, as well as reasonable lodging and meals at a rate not greater than the County per diem amount, and as authorized by the County reimbursement policy, **only** when the medical examination has been at the request of the Association. Mileage reimbursement will be made at the County mileage rate after the member has attended the appointment, for the number of miles from the member's residence directly to the examination location, as determined by Association staff from a mapping service on the Internet, rounded to the nearest whole mile. Reimbursement for meal costs may also include those for a companion where one is needed to transport the applicant to the site of the medical examination, but only when approved in advance by the Association.
- c. If the applicant fails without sufficient excuse to attend an examination scheduled at the request of the Association, the applicant shall reimburse the Association for any missed appointment fee charged by the examining physician. The determination of what constitutes a sufficient excuse shall be within the sole discretion of the Board. Failure to pay the reimbursement in full, or to make acceptable arrangements to pay it in installments, shall constitute cause to dismiss the application, with or without prejudice as determined by the Board.
- d. All costs incurred for any and all medical examinations, medical reports and other charges which are undertaken or prepared at **the request of the member** shall be borne solely by the member.

3.4. Other Investigation

The Association may hire third parties, including but not limited to a private investigator, to investigate the issues and/or claims made by a disability retirement applicant that relate to his/her entitlement to receive disability benefits.

3.5. Information from Employer

At any time during the review process, the Association may request that the applicant's employer provide information regarding the member's job duties, illness or injury, accommodations, or other pertinent data. Additional reasonable requests may be made at any time as deemed necessary by the Association. Requests shall include a release by the applicant or other appropriate documentation. A participating employer shall provide the underlying data, or a response if requested by the Association, in writing, no later than 30 days after the date of the request. The Retirement Administrator may grant an extension of time of up to 10 additional days to respond upon a showing of good cause by the employer.

3.6. Submittal to Board

After obtaining all information deemed to be relevant and necessary, staff shall submit the matter to the Board.

4. INITIAL BOARD DECISION

4.1. Advice from County Health Officer

As provided in Government Code section 31530, the Board may seek advice from the county health officer at any time, and, if requested by the Board, the county health officer shall attend the Board's meetings.

4.2. Board's Options

Upon review of an application for disability retirement, the Board may do any of the following:

- a. Grant the benefit applied for.
- b. If the applicant has applied for service-connected disability retirement, and is eligible for non-service-connected disability retirement, grant a non-service-connected disability retirement, and:
 - 1) Deny the application for service-connected disability retirement, and direct staff to inform the applicant of his/her right to request a hearing on the issue of service connection, or
 - 2) Refer the issue of service connection to a hearing for a determination.
- c. Refer the entire matter to a hearing for a determination on any or all issues presented.
- d. Deny the application in its entirety and direct staff to inform the applicant of his/her right to request a hearing.
- e. Refer the matter back to staff with instructions.
- f. Dismiss the application without prejudice.
- g. Dismiss the application with prejudice.

4.3. Board Decision

Each decision shall be made by and shall be the decision of the Board. Each such decision shall be based upon all of the information presented to the Board.

In the event the Board is unable to act, in whole or in part, due to lack of a majority vote in accordance with the Board's bylaws, the minutes shall indicate that pursuant to these Procedures, the inability to take action is deemed an action to bring the matter back at the next regular meeting. If the Board is still unable to act at the second meeting, the minutes shall indicate that the inability to take action is deemed an action to deny the application, in whole or in part as applicable, pursuant to these Procedures.

PART III.

HEARING PROCESS AND FINAL BOARD DECISION

5. PROCEDURES APPLICABLE TO ALL HEARINGS

5.1. Referral by Board to Hearing for Determination

- a. If the Board's decision is to refer the matter to a hearing for a recommendation, the Board shall assign the matter to a hearing officer. The applicant and the applicant's employer shall be notified within ten (10) days of the Board's action. The Association's counsel shall be concurrently notified by copy of said notice.
- b. The notice shall advise that the matter will be determined by a hearing on the record, unless a formal hearing is requested by any party.
- c. A written request for a formal hearing may be filed by any party within ten (10) days after the date of the notice. (See section 5.2.c. regarding requirements for requesting a formal hearing; see also section 1.11 regarding no extension of time if a document is mailed.)

5.2. Request for Hearing after Denial of Application by Board

- a. If the Board's decision is to deny the application in whole or in part, the member, and the member's employer, shall be notified within ten (10) days of the Board's decision. The applicant and the Association shall have the right to request a hearing. The party requesting a hearing may request either a hearing on the record, as provided in Article 6, or a formal hearing, as provided in Article 7. A formal hearing will be scheduled if requested by any party.
- b. A request for hearing **must** be filed with TCERA not later than ten (10) days after the date of the notice of denial. (See section 1.11 regarding no extension of time if a document is mailed.)
- c. If the request is for a formal hearing, the party **must** indicate on the included calendar any dates which are unacceptable for a hearing. The party **must** also indicate the expected number of hearing days required to hear the matter. When a

request for a formal hearing has been received, the Association shall immediately forward a copy to the non-requesting party, who shall within ten (10) days after the date of the notice indicate on the included hearing calendar any unacceptable dates. The parties are responsible for clearing calendars for any medical or other witnesses prior to the time that the responsive calendar is submitted. See section 7.3.c. (disallowed grounds for postponement or continuance).

If the applicant desires legal counsel, it is the applicant's responsibility to obtain counsel sufficiently in advance of the hearing date to allow counsel adequate time to prepare for the hearing. See section 7.3.c (disallowed grounds for postponement or continuance).

A failure of any party to submit a calendar indicating the dates not acceptable for hearing shall be deemed consent to set the hearing on any weekday indicated to be available on the calendars that were sent to both parties.

5.3. Changing Type of Hearing after Initial Selection

- a. A party who has requested a hearing on the record may not request that the matter be changed to a formal hearing.
- b. A party who has requested a formal hearing may request changing to a hearing on the record. Provided the other party has not requested a formal hearing, and/or stipulates to changing to a hearing on the record, the Retirement Administrator may grant the request.

The request must be filed with the Association not later than 30 days before the date the formal hearing is initially scheduled to begin.

If the Retirement Administrator grants the request, the triggering event for purposes of the 30-day period to file briefs as provided in section 6.2 shall be either the date of service of the approval, or the date set forth in section 6.2, whichever is later.

5.4. Assignment of Hearing Officer

- a. Once a hearing has been requested or ordered, the Association shall assign the next available hearing officer in the rotation to the hearing. The Association shall notify the parties promptly when a hearing officer has been assigned.
- b. Each party shall be entitled to request reassignment of the hearing to another hearing officer without cause. Each party shall be entitled to make only one (1) such request.
 - 1) The request must be in writing and must be filed with the Association no later than 10 days after service of the notice of assignment of hearing officer.

- 2) A party which did not request reassignment the first time may file its own request for reassignment as described above after the notice of assignment of the second hearing officer is served.
- c. At any time, a party may request reassignment to another hearing officer for good cause.
 - 1) The reassignment request shall be supported by a declaration under penalty of perjury, and shall be made in the form provided in Appendix A.
 - 2) Upon the filing of a request for reassignment in accordance with the provisions of this section 5.4.c., without any further act or proof, the Association shall reassign the case to the next available hearing officer in the rotation.
 - 3) Upon reassignment, a new notice of assignment of hearing officer shall be served on the parties. If there are no available hearing officers left in the rotation, the Association shall take such steps as are appropriate under the circumstances, which may include contracting with the Office of Administrative Hearings to provide an administrative law judge.

5.5. Authority of Hearing Officer

The hearing officer may require the parties to hold teleconferences as needed.

The hearing officer may also make any procedural rulings needed, and, upon the stipulation of all parties, the hearing officer may waive or modify any provision of the rules in Article 6 or 7 for that type of hearing, in whole or in part, except for the time limits in section 7.3.

The hearing officer shall not engage in ex parte communications with a party. Discussions with one party about scheduling the hearing do not constitute ex parte communications.

5.6. Initial Exhibit Packet

For all hearings, the Association shall prepare the initial exhibit packet as follows.

a. Assembly of Packet

Within 45 days after the notice of initial assignment of the hearing officer, the Association shall prepare and serve a consecutively numbered, tabbed exhibit packet consisting of all records provided by the applicant or the applicant's representative and/or the employer or employer's representative, and all records obtained by the Association in the application process. The initial exhibit packet shall consist only of those records in the Association's possession as of the date of the initial Board decision.

b. Service of Packet

The Association shall provide electronic copies of the initial exhibit packet to all parties, with a proof of service indicating the date of distribution. The Association shall also provide hard copies for the hearing officer and TCERA's counsel at no charge, and shall provide a hard copy for the applicant on request upon payment of the fee for preparation of a hard copy.

5.7. Supplemental Exhibit Packet

The Association shall provide a supplemental exhibit packet (if any), or a notice of no supplemental exhibit packet, as follows:

- a. Within 45 days after the date of service of the initial exhibit packet, a party shall do one of the following:
 - 1) Advise the Association in writing that the party will not submit any additional records for a supplemental exhibit packet.
 - 2) Advise the Association in writing that the party may wish to submit additional records for a supplemental exhibit packet.

This notice may include a written request to the Association to issue subpoenas for records to be provided directly to the Association, in circumstances where the party is legally barred from obtaining the records, or where the party shows the Association that reasonable efforts to obtain the records have been unsuccessful.

Neither notice needs to be served on the other party or parties.

- b. If a party wishes to submit additional records for a supplemental exhibit packet, that party must arrange for the additional records, whether subpoenaed or not, to be provided directly to the Association not later than 135 days after the date of service of the initial exhibit packet as described in section 5.6.b. (i.e., not later than 90 days after the deadline in section 5.7.a.).
- c. If all parties advise pursuant to section 5.7.a.1. that no additional records will be submitted, then, within 15 days after the date of receipt of the last notice, or the date in section 5.7.a., whichever is earlier, the Association shall notify the parties, with a proof of service, that no supplemental exhibit packet will be provided.
- d. If one or more parties advise the Association that additional records may be submitted, then within 15 days after the deadline in section 5.7.b., the Association shall prepare a consecutively numbered, tabbed supplemental exhibit packet consisting of all records provided by the deadline to the Association, and shall provide it to all parties and the hearing officer, with a proof of service indicating the date of distribution, in the same manner as in section 5.6.b.

5.8. Objections to Records in Packets

All records in the initial exhibit packet and the supplemental exhibit packet, if any, shall be deemed to be properly authenticated, and shall not be treated as hearsay unless the objecting party, not later than thirty (30) days after delivery of the initial or supplemental exhibit packet, files with the hearing officer, with a proof of service on the other party or parties, a request to exclude certain records, with a description detailing the basis for the objection with sufficient specificity. Nothing in this rule precludes a party from arguing about the substance of, or challenging the truth of, any material contained in the exhibit packet or packets as prepared by the Association. The rules of privilege shall be effective to the extent that they are otherwise required by statute to be recognized at the hearing.

5.9. Submittal of Additional Records

- a. No additional records may be submitted to the Association after the date of service of the supplemental exhibit packet, or notice of no supplemental exhibit packet, as described in section 5.7.
- b. The hearing officer may allow additional medical reports or other documents to be submitted directly to the hearing officer after the date in section 5.7. The requesting party must show that such records are directly relevant to an issue in dispute, and that the evidence is not cumulative. The hearing officer must permit submission in any of the following circumstances:
 - 1) Where records that were *timely* subpoenaed were not received by the Association in time to be included in the supplemental exhibit packet.
 - 2) Where records *which predate the date in section 5.7.a.* are discovered which the party neither knew about nor should have known about, in the exercise of reasonable diligence, and where the party proposing to submit such records provides notice to the other party or parties within 24 hours after learning of the existence of such records, and provides copies to the other party or all other parties within 24 hours after obtaining the additional records.
 - 3) With the consent of the Retirement Administrator.
- c. If the hearing officer permits submission under section 5.9.b., the party shall provide copies at no cost to the hearing officer and all other parties. The hearing officer may determine timelines for submission of documents not covered by section 5.9.b.2., and/or may set or reset other deadlines in Article 6 or 7 as needed, except those in section 7.3.
- d. In a hearing on the record, the hearing officer must reject any documents which do not meet the criteria in section 5.9.b.

e. In a formal hearing, the hearing officer may permit submission of documents which do not meet the criteria in section 5.9.b. See section 7.12 (Submittal of Additional Records at Hearing).

5.10. Additional Facts Arising Before Hearing

If facts which could have affected the Board's initial decision, but were not known at the time of that decision, come to light before the hearing commences, any party, or staff, may bring these facts to the attention of the Retirement Administrator. The Retirement Administrator may determine that these facts are material enough that they should be presented to the Retirement Board.

If that determination is made, the Retirement Administrator shall cancel any hearing process and shall return the matter to the Board for further review pursuant to Article 4 (Initial Board Decision). If the Board again denies the application, in whole or in part, the applicant may again request a hearing.

If a new request for hearing is submitted, the matter will be heard by the same hearing officer as assigned earlier, unless he/she is not available, in which case all procedures in section 5.4 (Assignment of Hearing Officer) will be followed.

6. HEARING ON THE RECORD

6.1. Exhibit Packet

The rules for the exhibit packet are set forth in Article 5. As noted in section 5.9.d., the hearing officer must reject any supplemental documents which do not meet the criteria set forth in section 5.9.b.

6.2. Briefs

Not later than thirty (30) days after the date of service of the supplemental exhibit packet, or notice of no supplemental exhibit packet, as described in section 5.7, the parties may file simultaneous written briefs (as defined in section 1.2.f.) with the Association, with a proof of service on the hearing officer and the other party or parties, which identifies issues, evidence, and relevant law, and presents arguments. (For help in writing a brief, parties may search the Internet for "how to write a legal argument in a court case.") If a brief claims that something is in the record, the brief must identify the volume and page of the record where the item appears.

One original of each brief must be filed in hard copy with the Association. The parties may stipulate to serve solely electronic copies on each other, and the hearing officer may consent to be served solely with an electronic copy.

The hearing officer may, for good cause, grant a request to extend time for the parties to file their simultaneous briefs. The request must be made no later than seven (7) days before

the original deadline, and the new deadline shall be no later than 15 days after the original deadline. The hearing officer shall notify the Association and the parties in writing whether the request for an extension has been granted or denied. If the request is granted, the deadline for the hearing officer's proposed findings of fact and recommendations pursuant to section 6.3 below shall be extended accordingly.

6.3. Hearing Officer's Recommended Decision

The hearing officer shall review the record and any briefs, and shall submit his/her proposed Findings of Fact and Recommendations to the Board of Retirement, with a proof of service on the parties, no later than 30 days after the date the briefs were filed or due, as described in section 6.2.

The hearing officer shall include in his/her recommended decision a notice to the parties that they have ten (10) days from the date of service of the recommended decision to submit written objections, with a proof of service on the other parties, to the Board.

6.4. Objections to Hearing Officer's Recommended Decision

Pursuant to Government Code section 31534, the parties shall have ten (10) days from the date of service of the hearing officer's proposed Findings of Fact and Recommendations to submit written objections, with a proof of service on the other parties, to the Board. (The objections are **not** served on the hearing officer.) These objections shall be incorporated in the record to be considered by the Board when it considers the proposed decision.

7. FORMAL HEARING

7.1. Initial Scheduling of Hearing

The hearing must initially be scheduled to commence within one hundred eighty (180) days after the date of service of the initial exhibit packet, as described in section 5.6.

The Association or its counsel shall initiate the scheduling process. The parties are expected to cooperate in setting the hearing date or dates. The hearing officer shall settle disputes regarding the initial scheduling of hearing dates.

The hearing may be postponed as provided in section 7.3.

Hearings that are not concluded on the date set may be continued to any future date agreed to by all parties, provided the new date meets the requirements of section 7.3.

7.2. Notice of Hearing

Once the initial hearing date has been established, the Association shall promptly issue a formal notice of hearing to each party describing the time and place of the hearing. Such notice shall be provided not later than forty-five (45) days prior to the initial hearing date.

7.3. Postponements and Continuances

a. Definitions

- 1) "Postponement" means delaying the beginning of a hearing prior to the submission of any formal evidence or testimony.
- 2) "Continuance" means delaying resumption of a hearing after the hearing has begun, i.e., after some formal evidence and/or testimony has been received from one or more of the parties.

b. Limitations

- 1) The hearing officer may grant one request by each party to postpone the hearing for good cause, and may also grant one request by each party to continue the hearing for good cause. Any request for postponement or continuance shall be made not later than fifteen (15) days in advance of the next scheduled date of the hearing, except when the basis for the postponement or continuance is not known fifteen (15) days in advance. The hearing officer shall promptly grant or deny any request.
- 2) Only the Retirement Administrator may grant additional requests by a party for postponement or continuance for good cause. The Retirement Administrator must also approve treating a postponement or continuance as not charged to any party, such as when the hearing officer requests a postponement or continuance on his/her own behalf.
- 3) Any postponement or continuance must be to a date certain which is within 45 days after the then currently scheduled date, or the most recent date on which the hearing was actually held, as applicable.
- 4) If the request is granted, the hearing officer shall indicate in writing whether the postponement or continuance is charged to a party, and, if yes, which party or parties have used their requests and which still have a request available.

c. Good cause

Good cause shall be within the discretion of the hearing officer or Retirement Administrator, as applicable. However, the following shall **not** constitute good cause for postponement or continuance of a hearing:

1) The desire to await the outcome of an application for workers' compensation or to exhaust benefits under Labor Code section 4850 and/or long term disability pay.

- 2) The need to permit a condition to stabilize in order to determine whether the disability is permanent.
- 3) Failure to obtain legal counsel, if so desired, sufficiently in advance of the hearing date to allow counsel adequate time to prepare for the hearing.
- 4) Conflicting schedule of a witness, where that conflict was known at the time the hearing calendar was submitted.

d. Hearing officer's authority

The hearing officer shall settle disputes regarding the scheduling of postponed or continued hearing dates.

7.4. No Discovery

No discovery requests to other parties are allowed, including depositions, requests for production of documents, requests for admission, and all other discovery permitted in civil cases, except as expressly permitted in these Procedures.

7.5. Submission of Witness Lists

Not later than thirty (30) days before the scheduled hearing date, all parties or their counsel shall submit to the hearing officer, with a proof of service on the other party or parties, a list of the names of any physicians or other witnesses who will or may be called to testify.

If the hearing is postponed or continued, the hearing officer may require a supplemental list to be submitted with such deadlines and other requirements as the hearing officer may set.

7.6. Independent Medical Examination

Not later than thirty (30) days before the scheduled hearing date, a party may request, or the parties may stipulate to request, that the Retirement Administrator authorize an independent medical examination. The request must be in writing and must show good cause to grant the request. The Retirement Administrator may grant the request if good cause is shown.

7.7. Subpoenas for Attendance of Witnesses at Hearing

It shall be the responsibility of each party to obtain those subpoenas for witnesses to appear at the hearing that he/she deems necessary for the presentation of his/her respective portion of the case. The party requesting the subpoenas must submit to the Association a written request with the name of each person to be subpoenaed. The parties are strongly advised to request subpoenas as soon as the party knows the witness will be called. The Association shall provide the completed subpoena forms within **five (5) working days** after such request. Subpoenas shall be signed by the Chair of the Board, the Retirement Administrator, or the hearing officer.

Neither the Board nor the Association is responsible for the service of subpoenas, the cost of service, or the payment of witness fees. All subpoenas are to be issued in accordance with Government Code section 31535. The parties and their counsel are responsible for complying with all legal requirements regarding the request for and service of subpoenas and regarding any other laws which may impact such subpoenas.

7.8. Affidavits

- a. At any time twenty (20) or more days prior to a hearing or a continued hearing, any party may mail or deliver to the opposing party a copy of any affidavit that he/she proposes to introduce, as set forth below. Unless the opposing party, within ten (10) days after such mailing or delivery, mails or delivers to the proponent a request to cross-examine an affiant, his/her right to cross-examine such affiant is waived and the affidavit, if introduced in evidence, shall be given the same effect as if the affiant had testified orally. If an opportunity to cross-examine an affiant is not afforded after request therefor is made as herein provided, the affidavit shall be excluded.
- b. The notice referred to in subdivision a. shall be substantially in the following form:

The accompanying affidavit of (here *insert name and contact information of affiant*) will be introduced as evidence at the hearing in (here *insert title of proceeding*). (Here *insert name of affiant*) will not be called to testify orally and you will not be entitled to question him/her unless you notify (here *insert name of proponent or his/her attorney*) at (here *insert address*) that you wish to cross-examine him/her. To be effective your request must be mailed or delivered to (here *insert name of proponent or his/her attorney*) on or before (here *insert a date 10 days after the date of mailing or delivering the affidavit to the opposing party*).

7.9. **Pre-Hearing Briefs**

A party may, but is not required to, submit a pre-hearing brief. (This is most often done when a case presents a technical, often legal, issue.) A party wishing to submit a prehearing brief shall notify the hearing officer and other parties no later than 14 days prior to the scheduled date of the hearing of the party's intent to submit a pre-hearing brief. One original of a pre-hearing brief must be filed in hard copy with the Association. Copies must be served on the hearing officer and the other party or parties in such a manner that they receive it no later than two (2) days prior to the hearing, unless otherwise ordered by the hearing officer. The other party or parties and the hearing officer may consent to be served solely with an electronic copy.

7.10. Stenographic Reporter

The Association shall arrange for a certified shorthand reporter to record the hearing at the expense of the Association.

7.11. Rights of Parties

Oral evidence shall be taken only on oath or affirmation. Each party shall have these rights: to call and examine witnesses; to request that additional records be accepted; to crossexamine opposing witnesses on any matter relevant to the issues even though that matter was not covered in the direct examination; to impeach any witness regardless of which party first called him/her to testify; and to rebut the evidence against him/her. If the applicant does not testify, he/she may be called and examined as if under cross-examination. At the request of any party, witnesses may be excluded from the hearing, except while testifying.

7.12. Submittal of Additional Records at Hearing

- a. The hearing officer may permit additional records to be submitted at the hearing. The requesting party must show why the records should be permitted despite not meeting the criteria in section 5.9.b.
- b. Should the hearing officer permit submission of records, he/she shall condition the submission upon such terms as he/she deems fair, including without limitation: permitting the non-submitting party to submit additional records; permitting the non-submitting party to subsequently file a declaration of other witnesses regarding the newly submitted evidence; permitting the non-submitting party to call the witness who authored the evidence at the expense of the submitting party; permitting the non-submitting party to depose the witness at the expense of the submitting party and have the transcript submitted as evidence; and/or continuing the hearing to permit time to properly respond to the new records, which, if granted, shall not be counted as the non-submitting party's one allowable request to the hearing officer. Additionally, the hearing officer may consider the records as offered but afford them only weight as uncorroborated hearsay.

Prior to ruling on the admissibility of any belatedly presented records, in order for the submitting party to determine if he/she wishes to pursue the issue, the hearing officer shall advise the submitting party of the possible conditions to admissibility which might be imposed. If the submitting party does pursue submission and the evidence is later admitted, it shall be conclusively presumed that the submitting party consents to the conditions identified by the hearing officer regarding the admission of the evidence.

c. A party wishing to submit additional documents shall bring enough copies to the hearing to assure that the hearing officer and all parties receive a copy.

d. The hearing officer shall ensure that any documents submitted are marked as additional exhibits, whether admitted into evidence or not, and shall forward the hearing officer's copies to the Association after the hearing for inclusion in the record as required by Code of Civil Procedure sec. 1094.6 ("Such record shall include ... all rejected exhibits in the possession of the local agency or its commission, board, officer, or agent").

7.13. Rules of Evidence

- a. Formal rules of evidence do not apply, but evidence may be admitted only if it is the kind of evidence upon which reasonable persons are accustomed to rely in the conduct of serious affairs. Hearsay evidence alone shall be insufficient to support a decision. As provided in section 5.8, records included in the exhibit packet(s) prepared by the Association are not hearsay unless a timely objection is made.
- b. Any objections to the admissibility or relevance of evidence other than evidence admitted pursuant to section 5.9.b. shall be determined by the hearing officer. Notwithstanding this rule, any party may object at the hearing that such documents are, in whole or in part, irrelevant.
- c. The hearing officer shall retain the right to exclude evidence which is not relevant or which is cumulative, or otherwise found not probative.
- d. Any party may request that the hearing officer take official notice of those matters provided for in Evidence Code sections 451 through 453.
- e. Other than rebuttal witnesses, neither party may call as a witness any physician or other witness whose name has not been disclosed to the other party pursuant to the provisions of section 7.5 (Submission of Witness Lists), except as provided in section 7.12.b. (possible conditions for submission of additional records at hearing).

7.14. No Additional Records after Conclusion of Hearing

No additional records may be submitted after the conclusion of the hearing. The hearing concludes when all testimony and other evidence has been received; any allowances for post-hearing briefs before the matter is deemed submitted to the hearing officer do not extend this time.

7.15. Transcript of Hearing

- a. The Association shall order a transcript of the hearing for the hearing officer's use. The costs shall be paid by the Association.
- b. If a copy of the transcript is ordered at the request of the applicant or the applicant's attorney, the costs shall be paid by the applicant.

c. The costs of a copy ordered by any party after the original transcript has been prepared shall be borne by the party ordering the copy.

7.16. Post-Hearing Briefs

Not later than thirty (30) days after the conclusion of the hearing, the parties may file simultaneous written briefs with the Association. The hearing officer may extend this deadline for good cause to 45 days. Alternatively, the parties may stipulate to, or the hearing officer may require, opening, opposing, and reply briefs, on a schedule to be determined by the parties, with the written description of the schedule to be provided promptly to the Association.

Any party which wishes to file a written brief shall file an original with the Association, with a proof of service on the hearing officer and the other party or parties, which sets forth the issues and contentions of the party and a summary of the evidence, and discusses applicable law. (For help in writing a brief, parties may search the Internet for "how to write a legal argument in a court case.") References to factual matters contained in the record must include a citation to the volume and page of the exhibits where this information is shown.

One original of each brief must be filed in hard copy with the Association. The parties may stipulate to serve solely electronic copies on each other, and the hearing officer may consent to be served solely with an electronic copy.

The matter shall be deemed submitted on the day the post-hearing briefs are, or the reply brief is, due. If the parties waive the filing of post-hearing briefs, the matter shall be deemed submitted at the conclusion of the hearing.

7.17. Failure to Pursue Application

If, as a result of the applicant's failure to comply with the procedures specified above, **including but not limited to failure to cooperate in scheduling the hearing, or unwarranted requests to postpone or continue a hearing,** the hearing is not concluded within nine (9) months after a request or referral for a formal hearing, the case shall be dismissed with prejudice. This remedy pertains to a matter that has already been sent to a hearing and does not supersede the dismissal without prejudice provided for in section 2.5 (time limit for filing application).

7.18. Findings of Fact and Recommendations

a. Not later than thirty (30) days after the date the matter is deemed submitted, as provided in section 7.16, the hearing officer shall serve his/her proposed Findings of Fact and Recommendations on the parties and file them with the Board. The hearing officer may file with the Association a written request for one 30-day extension for submission of the recommended decision, which request shall be granted or denied by the Retirement Administrator.

One original of the proposed Findings of Fact and Recommendations must be filed in hard copy with the Association. The parties may stipulate to allow the hearing officer to serve solely an electronic copy on them.

The hearing officer shall include in his/her recommended decision a notice to the parties that they have ten (10) days from the date of service of the recommended decision to submit written objections, with a proof of service on the other parties, to the Board.

b. Pursuant to Government Code section 31534, the parties shall have ten (10) days from the date of service of the hearing officer's proposed Findings of Fact and Recommendations to file written objections, with a proof of service on the other party or parties, with the Association. (The objections are **not** served on the hearing officer.) These objections shall be incorporated in the record to be considered by the Board when it reviews the proposed decision.

One original of any objections must be filed in hard copy with the Association. The party receiving any objections may consent to be served solely with an electronic copy.

7.19. Shortening Time Periods

Nothing in these Procedures is to be construed as preventing the parties from stipulating to lesser intervals than those prescribed above. The hearing officer may also, for good cause shown, after giving both parties an opportunity to be heard, shorten the times specified in this Article. (See also section 5.5 regarding hearing officer's authority to waive or modify some procedures.)

8. BOARD DECISION AFTER HEARING

8.1. Board's Options

Upon receipt of the hearing officer's proposed Findings of Fact and Recommendations and considering the record, including any objections submitted by the parties, the Board may, in accordance with Government Code section 31534:

- a. Approve and adopt the proposed findings and the recommendations of the hearing officer;
- b. Require a transcript, plus all other evidence received by the hearing officer and, upon the receipt thereof, take such action as in its opinion is indicated by such evidence;
- c. Refer the matter back with or without instructions to the hearing officer for further proceedings; or

d. Set the matter for hearing before itself. At such hearing the Board shall hear and decide the matter as if it had not been referred to the hearing officer.

8.2. Referral Back to Hearing Officer

If the Board refers the matter back to the hearing officer with a request to amend the proposed decision, the hearing officer shall submit the revised proposed decision within 30 days after the date of the request from the Board.

If the Board refers the matter back to the hearing officer with instructions for further proceedings, the Board shall include in its instructions a deadline for completion of the additional proceedings.

8.3. Findings and Conclusions by the Board

In any case where the Board sets the matter for hearing before itself, the Board shall adopt such findings of fact and conclusions of law as it deems appropriate.

8.4. Notice of Decision

The Association shall give written notice of the Board's decision to all parties and their representatives within ten (10) days. Said notice shall include the decision and the date it was rendered.

8.5. Finality of Decision

Except as otherwise provided in Article 10 (judicial review), action by the Board granting or denying in whole or in part any application for disability retirement shall be deemed to be a final and conclusive determination of any and all issues raised by the application which were either presented to the Board or could with the exercise of reasonable diligence have been presented to the Board. No employee whose application for disability retirement has been denied by the Board, and who thereafter returns to service with a participating employer, shall be precluded from filing a new application which asserts as a basis for the application circumstances existing or known at the time of denial of the first application, in combination with circumstances which have arisen or become known after denial of the first application. Such new application shall not, however, relate back to the date of an original application that was denied, and any entitlement to benefits shall be effective as of the date of filing of the new application.

PART IV.

MISCELLANEOUS

9. CHANGE OF POSITION IN LIEU OF DISABILITY RETIREMENT ALLOWANCE

As required by Government Code sections 31725.5 and/or 31725.65, if a disability applicant is found disabled and returns to County or other participating employer service in a position which pays less than the position from which he/she became disabled, the Association shall pay, in lieu of a disability retirement allowance, the difference between the member's former salary and current salary. Members should also note that if a safety member is placed in a general position, the member retains safety status.

10. JUDICIAL REVIEW

10.1. Judicial Review

In those cases where a party is entitled to judicial review of the proceedings, the petition to the court shall be filed in accordance with Code of Civil Procedure section 1094.5. Notice of the right to review under section 1094.5, and the time limits for filing for such review under Code of Civil Procedure section 1094.6, shall accompany the written notice of the Board's decision.

If the member's application was denied on the ground that the medical examination and other available information did not show to the satisfaction of the Board that the member was incapacitated physically or mentally for the performance of his/her duties, the notice to the employer shall advise the employer that the employer's time limit for obtaining judicial review is governed by Government Code section 31725, which requires the employer to obtain judicial review of the Board's action by filing a petition for writ of mandate within 30 days after the mailing of the notice. If the employer's application was denied on this ground, the notice shall advise the employer that this deadline **may** apply.

10.2. Preparation of Administrative Record

Pursuant to Code of Civil Procedure section 1094.6, TCERA staff shall prepare the administrative record for court upon written request by the petitioner. The petitioner must pay the cost of preparing the original of the administrative record in hard copy for the court. The petitioner must pay the estimated cost of preparation before TCERA prepares the administrative record. If the litigation is abandoned or dismissed before TCERA has completed preparation of the administrative record, or if the deposit exceeds the preparation cost, TCERA will refund any portion of the deposit exceeding the preparation cost actually incurred.

Pursuant to Code of Civil Procedure section 1094.5, the petitioner may ask the court to find that the petitioner cannot afford to pay any court costs, including the cost of the administrative record. If the court issues an order finding the petitioner to be indigent, and the petitioner provides a copy

of the signed order to TCERA, TCERA shall then pay the cost of providing the original of the administrative record in hard copy for the court. (This cost is recoverable as court costs if TCERA prevails in the litigation.) TCERA shall file the original with the court.

TCERA will provide a hard copy of the administrative record to its counsel at no charge. Any party other than TCERA must pay a fee to obtain a copy, and may choose what type of copy to receive: only an electronic copy (minimal fee), only a hard copy (higher fee), or both. If the petitioner obtains a court order of indigency, the procedure set forth above for the original of the administrative record (which is submitted to the court) shall also apply to the copy for the petitioner.

11. RE-EXAMINATIONS

The Board may, pursuant to Government Code section 31729, require any disability retiree under age 55 to undergo medical examination. The examination shall be made by a physician appointed by the Board. Upon the basis of the examination the Board shall determine whether the disability retiree is still physically or mentally incapacitated for service in the job classification held by him/her when retired for disability. The Board may additionally, or in the alternative, arrange for an investigation of the circumstances pertaining to such retiree that relate to his/her entitlement to receive disability benefits.

In the event that the examination or investigation shows that the retiree may not be incapacitated, the Board shall thereupon notify the retiree, at his/her last known address, of said recommendation of the medical advisor and advise the retiree that the Board will meet to reconsider his/her disability retirement status. If the Board's decision is to refer the matter to a hearing officer for a determination, the rules as set forth in Articles 5 through 7 (prehearing procedures, hearing on the record procedures, and formal hearing procedures), to the extent applicable, shall apply. Said notice shall inform the retiree of his/her right to request a formal hearing, provided said request is submitted in writing no later than fifteen (15) days after the date of the notice.

12. SUBROGATION

Disability applicants are hereby notified that the 1937 Act provides for subrogation of benefits. (See Gov. Code, § 31820 et seq.) This means that if the Association is paying benefits in a situation where someone other than the employer is responsible for the injury, the Association may seek to obtain partial reimbursement from that person of the benefits paid. In doing so, the Association may join with the employer or the employer's workers' compensation carrier in any proceeding, or may file its own action or a claim in an action filed by the employee.

PART V.

EMPLOYER-FILED APPLICATIONS

13. EMPLOYER-FILED APPLICATION

13.1. Applicability

The provisions of this Article shall be in accordance with Government Code section 31721. This statute provides that an employer may not separate because of disability a member otherwise eligible to retire for disability, but shall apply for disability retirement of any eligible member believed to be disabled. The Association shall provide appropriate information to the employer regarding eligibility of a member for disability retirement upon request by the employer.

All provisions of these Procedures shall be applicable, substituting the employer-applicant for the member-applicant, except as provided below.

13.2. Application

The application for disability retirement shall consist of each and all of the following:

- a. A completed Application for Disability Retirement;
- b. A written statement signed by the department head or his/her authorized agent which asserts that a member of the Association is disabled in a manner which permanently incapacitates the member from the substantial performance of his/her duties, describes the nature and extent of the disability, and identifies the effect which the disability has had upon performance by the member of his/her duties; and
- c. A written statement which is as close as the employer can come to the equivalent of the treating physician's statement, which indicates that the member is permanently incapacitated for the performance of his/her duties.
 - 1) If the employer has independent subpoena power (such as the Board of Supervisors' power under Gov. Code, § 25170 et seq.), and the employer is unable to obtain a written statement from the member's treating physician, this statement shall consist of, at a minimum, a written opinion from a physician who has reviewed the medical records obtained by the employer pursuant to its subpoena power.
 - 2) If the employer does not have independent subpoena power, and the employer is unable to obtain a written statement from the member's treating physician, the Association may accept the application without complete information, provided the written statement by the employer sufficiently

shows why the employer believes the member is permanently incapacitated, or may reject the application until the employer provides more information supporting its belief that the member is permanently incapacitated. If the Association accepts the application, the Association shall then (notwithstanding section 7.4 prohibiting discovery during the hearing stage) provide to the employer such deposition and other_subpoenas as requested by the employer during the investigation stage, so that the employer may have medical records sent to TCERA, and may then make arrangements with TCERA to have a physician review the medical records and provide a written opinion based on the records to the Association regarding whether the member is permanently incapacitated, and if yes whether on a serviceconnected basis or not.

d. The application will be assumed to be on a non-service basis unless the employer affirmatively states that the employer believes the incapacity is service-connected. As set forth in section 13.3.d., the member may later assert that the incapacity is service-connected.

13.3. Notice

The Association shall, not later than ten (10) days after the date of the application is filed, personally, or by certified mail, serve on the member the following:

- a. A copy of these Procedures.
- b. A copy of the employer's application.
- c. The Disability Retirement Questionnaire and the release forms to enable the Association to obtain the Applicant's medical and personnel records.
- d. A written notice which informs the member:
 - 1) Of the application and the date of its filing;
 - 2) Of the member's right to either support or oppose the application and to be represented by legal counsel at the member's expense;
 - 3) Of the member's right to provide one or more medical reports pertaining to the application to the Association, at the member's expense.
 - 4) Of the member's right to amend the application to become the applicant, and that, if the member does so, the member may also amend the application for non-service-connected disability retirement to apply for serviceconnected disability retirement, or that, alternatively, the member may file **a new application** for service-connected retirement after the employer has shown that the member is permanently incapacitated.

- 5) Of the requirement that, if the member becomes the applicant, the member shall provide all information required as described in Article 2.
- 6) That, unless the member becomes the applicant, the employer shall have sole control of any decisions normally made by the applicant.
- 7) That any communications between the Association and the member shall not be deemed to substitute for any communications between the employer and the member.

13.4. Burden of Proof

Where the employer has filed the application, the employer has the burden of showing permanent incapacity. However, as provided in section 13.3.d.4., the member may either amend an employer-filed application to become the applicant, or wait and file a new application. The burden of proof and the procedures shall therefore be as follows.

- a. Unless and until the member amends the application to become the applicant, the employer shall have the burden of proof to show permanent incapacity and, if applicable, service-connectedness.
- b. If the member amends the application to become the applicant, the member shall have the burden of proof to show permanent incapacity and, if applicable, service-connectedness.
- c. The member's right to amend the application to become the applicant shall expire when the Board makes its initial decision as described in section 13.7 below. After that date, the member must file a new application for any type of disability not granted by the Board in the initial decision, within the time provided in this Article.

13.5. Costs

Even if the employer is the applicant, all physician, medical, and other costs incurred for any and all examinations and medical reports undertaken or prepared **at the request of the member** shall be borne solely by the member. The employer and member may agree that the costs of any examination or medical report will be paid by the employer.

13.6. Review of Application

Review of an employer-filed application shall be in the same manner as a member-filed application, except that where the member is not cooperating in the review process, the Association will not schedule a medical examination to be paid for by the Association.

13.7. Initial Decision of Board Where Member Has Not Amended Employer's Application to Become Applicant

The initial decision by the Board shall be in accordance with Article 4, except that the actions described below shall result in the following procedures:

a. Decision to grant employer's application for non-service connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board. The notice to the member shall include information regarding the member's right to request a hearing in the event the employee is in disagreement with the decision of the Board. The employee must file the request for hearing within 10 days from the date of the notice, in the same manner as where the member is the applicant, except that the hearing shall not consider the issue of service connection.
- 2) At the close of the 10-day period for the member to request a hearing, the Association shall notify the employer whether the member has filed a request for hearing regarding the decision of the Board.
- 3) If the member does not file a request for hearing, the Board's decision as to permanent incapacity shall be final. The member shall then be sent a notice stating that the member has 10 days from the date of the notice to file a **new** application for service-connected disability retirement.
- 4) At the close of this second 10-day period, the Association shall notify the employer whether the member has filed a new application for service-connected disability retirement. If no new application is filed, the Association shall notify the parties that the matter is final.

b. Decision to <u>deny</u> employer's application for non-service-connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board.
- 2) The notice shall include information regarding the employer's right to request a hearing in the event that the employer is in disagreement with the decision of the Board. The notice to the employer shall include the information that failure to request a hearing shall result in the member being reinstated to his/her employment, in accordance with Government Code section 31725. The employer must file the request for hearing within 10 days after the date of the notice, in the same manner as where the member is the applicant, except that the hearing shall not consider the issue of service connection. A member who has not amended the application to become the applicant may not request a hearing.

- 3) If the employer files a request for hearing, the matter shall proceed as described in Articles 5-7.
- 4) If the employer does not request a hearing, the Board's decision shall be final, and the Association shall so notify the employer and member.
- 5) If the employer originally requests a hearing, and then later requests to withdraw the application before the hearing, the hearing will be vacated and the decision denying the application shall be final. (A **new** application may be filed thereafter by any party.)

c. Decision to <u>grant</u> employer's application for service-connected disability retirement

- 1) The parties shall be notified within 10 days of the decision of the Board. The notice to the member shall include information regarding the member's right to request a hearing in the event the member is in disagreement with the decision of the Board. The employee must file the request for hearing within 10 days after the date of the notice, in the same manner as where the member is the applicant.
- 2) At the close of the 10-day period for the member to request a hearing, the Association shall notify the employer whether the member has requested a hearing regarding the decision of the Board.
- 3) If the member does not file a request for hearing, the Board's decision shall be final, and the Association shall so notify the parties.

d. Decision to <u>grant</u> non-service-connected disability retirement, but to deny employer's application as to service connection

- 1) The employer and member shall be notified within 10 days of the decision of the Board. The notice shall inform the employer that the employer may request a hearing on the question of service connection, and shall inform the member that the member may request a hearing on the question of permanent incapacity. Either request must be filed within 10 days from the date of the notice.
- 2) If any party files a request for hearing, the matter shall proceed as described in Articles 5-7, except that if the member is the only party who requests a hearing, the hearing shall not include the issue of service connection. If one party requests a formal hearing and the other party requests a hearing on the record, the procedures for a formal hearing shall apply.

- 3) If no party requests a hearing, the Board's decision shall be final, and the Association shall so notify the parties.
- 4) If the employer requests a hearing, but later withdraws the request for hearing, the member shall then be sent a notice stating that the member's request for hearing shall continue, in the form of hearing requested by the member, but only on the question of permanent incapacity, and that if the member wishes to reinstate the issue of service connection, the member has 10 days to file a **new** application for service-connected disability retirement.

e. Decision to <u>deny</u> employer's application for service-connected disability retirement

- 1) The employer and member shall be notified within 10 days of the decision of the Board. The notice shall include information regarding the employer's right to request a hearing in the event that the employer is in disagreement with the decision of the Board. The notice to the employer shall include the information that failure to request a hearing shall result in the member being reinstated to his/her employment effective as of the day following the effective date of dismissal, in accordance with Government Code section 31725. The employer must file a request for hearing within 10 days from the date of the notice.
- 2) If the employer requests a hearing, the matter shall proceed as described in Articles 5-7.
- 3) If the employer does not request a hearing, the Board's decision shall be final.
- 4) If the employer originally requests a hearing, and then later withdraws the request for hearing, the hearing will be vacated, and the decision denying the application shall be final. (A new application may be filed thereafter by any party, with a new effective date if granted.)

f. Decision to dismiss application without prejudice

- 1) The employer and member shall be notified within10 days of the decision of the Board. The notice shall provide information regarding the employer's and member's rights to re-apply for a disability retirement.
- 2) Should the employer or member re-apply for a disability retirement, the effective date of disability retirement, if granted, will relate to the date of the re-application as required by statute.

13.8. Initial Decision of Board Where Member Has Amended Employer's Application to Become Applicant

Where the employer has applied for disability retirement, but the member has amended the application to become the applicant, with or without adding a request for service-connected disability retirement, if applicable, the initial decision of the Board shall be in accordance with Article 4.

13.9. Hearing Procedure

The hearing procedure shall be the same for an employer-filed application as for an application filed by a member, except that the member and his/her legal counsel shall also have the opportunity to participate, and shall have all the same rights and duties as the employer-applicant and its legal counsel and the Association and its legal counsel.

The hearing officer may make any rulings and/or modify the hearing process, except for the time limits provided in section 7.3, as needed to provide a fair hearing involving a third party.

13.10. Board Decision after Hearing

Where the employer remains the applicant, the decision of the Board after a hearing shall generally be made as provided in Article 8, with "employer" substituted for "member" where appropriate.

However, where the employer applied only for non-service-connected disability retirement, and the Board approves a recommendation by the hearing officer to grant the application, the initial notice of the Board's decision under section 8.3 shall not state that the decision of the Board is final. Instead, the notice shall state that the member has 10 days to file a **new** application for service-connected disability retirement.

If no new application for service-connected disability retirement is filed, at the close of this 10-day period the Association shall notify the parties that the matter is final.

If a new application for service-connected disability retirement is filed, the Association shall inform the employer of this fact, and shall advise the parties that the matter will not be final until the Board reaches a decision on the question of service connection.

The new application for service-connected disability retirement shall be deemed timely if the employer's application was filed within the time limits in Government Code section 31722, as described in section 2.5 of these Procedures. The new application shall be processed and decided upon as provided in these Procedures, except that the additional materials normally required to be included with an application by a member need not be provided at the same time, but instead may be provided within 30 days after the application is filed.

13.11. Judicial Review

The opportunity to obtain judicial review shall be as set forth in Article 9.

BBG/20171017/3/26/20/

APPENDIX A

STATE OF CALIFORNIA) COUNTY OF TULARE)

I, ______, hereby declare that I am the applicant (attorney for the applicant) (attorney for the Association regarding the application of ______) for a disability retirement pension. I believe that I (the applicant) (the Association) cannot have a fair and impartial hearing before the hearing officer to whom the case is assigned.

I declare under penalty of perjury that the foregoing is true and correct.

Executed on this _____ day of _____, 20__, at _____, California.

Signature